# STRADLEY RONON Fiduciary Governance Group

fiduciarygovernanceblog.com



The Practical Effects of Regulation Best Interest, Form CRS, and Advisers Act Interpretations on Broker-Dealers, Investment Advisers and Investment Companies

# Fiduciary Governance Group

fiduciarygovernanceblog.com

# **Table of Contents**

### Page

SEC Adopts Regulation Best Interest, Form CRS, and Advisers Act Interpretations	4
A Guide to Regulation Best Interest	.15
SEC Issues Interpretive Release on Investment Adviser Standard of Conduct	42
SEC Adopts Form CRS Relationship Summary Requirement	49
Authors	58

# SEC Adopts Regulation Best Interest, Form CRS, and Advisers Act Interpretations



# **Fiduciary Governance**

June 6, 2019

### **Risk&Reward** fiduciarygovernanceblog.com

### SEC Adopts Regulation Best Interest, Form CRS, and Advisers Act Interpretations



The Securities and Exchange Commission ("SEC") has adopted a package of rules and interpretations governing the standards of conduct of broker-dealers and investment advisers and their associated persons, along with related disclosure requirements. The controversial actions were adopted on June 5, 2019, by a vote of 3 – 1, with Commissioner Robert Jackson dissenting on each vote. The rulemaking package consists of four separate actions:

- Regulation Best Interest ("Regulation BI") establishes a code of conduct for broker-dealers when making recommendations to retail customers."
- Registered broker-dealers and registered investment advisers will be required to provide a standardized relationship summary to retail investors on Form CRS.<sup>2</sup>
- The SEC published an interpretation of the standard of conduct for investment advisers under the Investment Advisers Act of 1940 ("Advisers Act").<sup>3</sup>
- The SEC also published an interpretation of Section 202(a)(11)(C) of the Advisers Act, which
  excludes from the definition of "investment adviser" any broker-dealer that provides advisory
  services when such services are "solely incidental" to the conduct of the broker-dealer's business.<sup>4</sup>

© 2019 Stradley Ronon Stevens & Young, LLP | www.stradley.com | Philadelphia | Washington | Chicago | New York

Information contained in this publication should not be construed as legal advice or opinion or as a substitute for the advice of counsel. The enclosed materials may have been abridged from other sources. They are provided for educational and informational purposes for the use of clients and others who may be interested in the subject matter.

### fiduciarygovernanceblog.com

This Client Alert provides our initial, high level summary of the SEC actions. We will be following up in the near future with a much more detailed analysis. In addition, as discussed below, we will be participating next month in a webinar that will examine the actions more thoroughly. In the meantime, please feel free to contact one or more of the individuals identified at the end of this Alert if you have specific comments or questions.

### Overview

At a high level, the SEC's final actions are consistent in content and scope with what it proposed.<sup>5</sup> Consistent with its approach in the proposing releases, the SEC ultimately chose not to impose a uniform fiduciary duty on broker-dealers and investment advisers. Instead, it elected to enhance the obligations of broker-dealers to bring them more closely in line with those of investment advisers. These enhanced obligations are derived in part from fiduciary principles, but the SEC did not characterize them as imposing actual fiduciary status on broker-dealers.

Despite the SEC's efforts to harmonize the obligations of broker-dealers and investment advisers, they will continue to be subject to different standards of conduct. Consequently, broker-dealers and investment advisers and their counsel will continue to have to consider and address the complexities associated with these different regulatory regimes.

Much of what is expected under the rules and forms, particularly Regulation BI, is not in the rules or forms themselves. Instead, important guidance is set forth in the releases. A thorough and detailed review of the SEC releases is critical to understanding the SEC actions and complying with them.

### **Regulation BI**

The most prominent element of the rulemaking package is Regulation BI, which is Rule 15I-1 under the Securities Exchange Act of 1934. This "regulation" (really just a single rule) enhances the existing standard of conduct applicable to broker-dealers (and their associated persons who are natural persons) at the time they recommend to a retail customer a securities transaction or investment strategy involving securities. Notably, this includes recommendations of account types and rollovers or transfers of assets (e.g., to roll over or transfer assets in a workplace retirement plan to an individual retirement account). "Recommendations" also include implicit hold recommendations resulting from agreed-upon account monitoring.

### General

Regulation BI requires that a broker-dealer, when making a recommendation, act in the retail customer's best interest and not place its own interests ahead of the customer's interests. This general obligation is satisfied only if the broker-dealer complies with four specified component obligations, referred to as the Disclosure Obligation, the Care Obligation, the Conflict of Interest Obligation, and the Compliance Obligation (collectively, the "Obligations").

A customer's "best interest" is not expressly defined. Whether a broker-dealer has acted in the customer's best interest will turn on an objective assessment of the facts and circumstances of how the specific components of Regulation BI, including the Obligations, are satisfied at the time that the recommendation is made (and not in hindsight).

### fiduciarygovernanceblog.com

The Regulation BI standard is not a fiduciary standard, but it includes key elements that are similar to key elements of the fiduciary standard for investment advisers. The SEC's intention is that, regardless of whether a retail investor chooses a broker-dealer or an investment adviser, the retail investor will be entitled to a recommendation or advice that is in the best interest of the retail investor and that does not place the interests of the firm or the financial professional ahead of the interests of the retail investor.

Regulation BI does not apply to advice provided by a broker-dealer that is dually registered as an investment adviser ("dual registrant") when acting in the capacity of an investment adviser. A dual registrant acts in the capacity of an investment adviser solely with respect to accounts for which the dual registrant provides advice and receives compensation that subjects it to the Advisers Act.

SEC Chairman Jay Clayton noted that he anticipates that FINRA will need to review and revise its rulebook and examination program in light of the enhanced broker-dealer standard of conduct reflected in Regulation BI.<sup>6</sup>

The Department of Labor has already indicated informally that it will pursue rulemakings and other guidance that "align" with Regulation BI. We expect that to occur this year, most likely in the Fall.

It is too early to tell whether these new rules will assuage the concerns of those states that have already introduced legislation or regulations aimed at governing broker-dealer and investment adviser standards of conduct (although the initial reaction is not promising). New Jersey and Nevada are the most notable examples to date.

### **Disclosure Obligation**

Under the Disclosure Obligation, before or at the time of the recommendation, a broker-dealer must disclose, in writing, all material facts about the scope and terms of its relationship with the customer. This includes a disclosure that the firm is acting in a broker-dealer capacity; the material fees and costs the customer will incur; and the type and scope of the services to be provided, including any material limitations on the recommendations that could be made to the retail customer.

The broker-dealer must also disclose all material facts relating to conflicts of interest that are associated with the recommendation. A conflict of interest is defined as an interest that might incline a broker-dealer or associated person – consciously or unconsciously – to make a recommendation that is not disinterested.

In connection with its proposal of Form CRS, the SEC initially proposed a rule specifically prohibiting the use of "adviser" or "advisor" by brokerage personnel not subject to the Advisers Act. The SEC did not adopt this rule, but it states that the use of these terms in a name or title by a broker-dealer (that is not also an investment adviser) or associated person (that is not also a supervised person of an investment adviser) would be a violation of the capacity disclosure requirement.<sup>7</sup> The SEC recognizes that there are exceptions to this general rule, such as for municipal advisors, commodity trading advisers, and advisors to special entities.

### **Care Obligation**

Under the Care Obligation, a broker-dealer must exercise reasonable diligence, care, and skill when making a recommendation to a retail customer. The broker-dealer must understand potential risks, rewards, and costs associated with the recommendation. The broker-dealer must then consider those risks, rewards, and costs in light of the customer's investment profile and have a reasonable basis to believe that the recommendation is in the customer's best interest and does not place the broker-dealer's interest ahead of the retail customer's interest. While costs must always be considered, they should be considered in light of other factors and the retail customer's investment profile; the standard does not necessarily require the "lowest cost option."

A broker-dealer should consider reasonable alternatives, if any, offered by the broker-dealer in determining whether it has a reasonable basis for making the recommendation. This does not require an evaluation of every possible alternative (including those offered outside the firm), nor does it require broker-dealers to recommend one "best" product. However, when a broker-dealer materially limits its product offerings to certain proprietary or other limited menus of products, it must still comply with the Care Obligation and cannot use its limited menu to justify recommending a product that does not satisfy the obligation to act in a retail customer's best interest.

When recommending a series of transactions, the broker-dealer must have a reasonable basis to believe that the transactions taken together are not excessive, even if each is in the customer's best interest when viewed in isolation.

### Conflict of Interest Obligation

Under the Conflict of Interest Obligation, a broker-dealer must establish, maintain, and enforce reasonably designed written policies and procedures addressing conflicts of interest associated with its recommendations to retail customers. These policies and procedures must be reasonably designed to identify all such conflicts and at a minimum disclose or eliminate them.

The policies and procedures must be reasonably designed to mitigate conflicts of interests that create an incentive for an associated person to place his or her interests or the interest of the firm ahead of the retail customer's interest. The SEC revised the mitigation requirement in the proposing release to focus on mitigating conflicts arising from associated persons' incentives and to eliminate the distinction between financial incentives and all other conflicts of interest.

In addition, when a broker-dealer places material limitations on recommendations (e.g., offering only proprietary or other limited range of products), the policies and procedures must be reasonably designed to disclose the limitations and associated conflicts and to prevent the limitations from causing the associated person or broker-dealer to place the person's or the firm's interests ahead of the customer's interest.

The policies and procedures must also be reasonably designed to identify and eliminate any sales contests, sales quotas, bonuses, and non-cash compensation (such as merchandise, gifts and prizes, travel expenses, meals and lodging) that are based on the sale of specific securities or specific types of securities within a limited period of time. The requirement is designed to eliminate such incentives when they create pressure (i) to sell a specifically identified type of security (ii) within a

### fiduciarygovernanceblog.com

limited period of time. Other incentives and practices that are not explicitly prohibited are permitted, provided that the broker-dealer establishes reasonably designed policies and procedures to disclose and mitigate the incentive created to the representative, and the Care and Disclosure Obligations are complied with.

### **Compliance Obligation**

Under the Compliance Obligation, a broker-dealer must establish, maintain, and enforce written policies and procedures reasonably designed to achieve compliance with Regulation BI as a whole. Thus, a broker-dealer's policies and procedures must address not only conflicts of interest but also compliance with its Disclosure and Care Obligations.

### Other Issues

The SEC will not have to show scienter (bad intent) to establish a violation of Regulation BI.

Although commenters urged the SEC to take a position that Regulation BI preempts or does not preempt state laws, the SEC chose not to do so. Instead, the SEC states that the preemptive effect of Regulation BI on any state law governing the relationship between regulated entities and their customers would be determined in future judicial proceedings based on the specific language and effect of that state law. In the open meeting, Chairman Clayton indicated that he hoped that states would work with the SEC going forward.

A broker-dealer will not be able to waive compliance with Regulation BI, nor can a retail customer agree to waive his or her protections under Regulation BI.

The SEC does not believe Regulation BI creates any new private right of action or right of rescission, nor does it intend such a result.

Compliance with Regulation BI will not alter a broker-dealer's obligations under the general antifraud provisions of the federal securities laws. Regulation BI applies in addition to any applicable securities laws and regulations.

### Form CRS Relationship Summary

The SEC has adopted rule and form amendments to require registered investment advisers and registered broker-dealers to provide a brief relationship summary to retail investors.<sup>8</sup> The relationship summary is intended to inform retail investors about (i) the types of client and customer relationships and services the firm offers; (ii) the fees, costs, conflicts of interest, and required standard of conduct associated with those relationships and services; (iii) whether the firm and its financial professionals currently have reportable legal or disciplinary history; and (iv) how to obtain additional information about the firm. The relationship summary will have a standardized questionand-answer format and must be no longer than two pages (four pages for dual registrants). Under certain standardized headings, firms will generally use their own wording to address the required topics. Firms are encouraged to use graphics, hyperlinks, and electronic formats. If a firm uses electronic formats, there are requirements for embedded hyperlinks to facilitate layered disclosure.

### fiduciarygovernanceblog.com

The relationship summary is designed to help retail investors select or determine whether to remain with a firm or financial professional by providing better transparency and summarizing in one place selected information about a particular broker-dealer or investment adviser. The format of the relationship summary is intended to allow for comparability among the two different types of firms in a way that is distinct from other required disclosures.

The relationship summary includes a required introductory paragraph that provides a link to Investor.gov/CRS, a page on the SEC's Office of Investor Education website that will offer educational information about investment professionals. This page currently summarizes the final release and promises future tailored information to educate retail investors about financial professionals, including tools to research firms and financial professionals, information about brokers and advisers, and information about brokers' and advisers' different services and fees. It also currently links to other resources that already were available on the Office of Investor Education website.

The relationship summary includes prescribed wording to describe the standard of conduct applicable to investment advisers and broker-dealers. In contrast to the proposed Form CRS, the standard of conduct is described as a "best interest" standard in all cases.<sup>9</sup>

As proposed, investment advisers must deliver a relationship summary to each new or prospective client who is a retail investor before or at the time of entering into an investment advisory contract with the retail investor. In a change from the proposal, broker-dealers must deliver the relationship summary to each new or prospective customer who is a retail investor before or at the earliest of (i) a recommendation of an account type, a securities transaction, or an investment strategy involving securities; (ii) placing an order for the retail investor; or (iii) the opening of a brokerage account for the retail investor. Broker-dealers and investment advisers must update the relationship summary and file it within 30 days whenever any information in it becomes materially inaccurate, and any changes must be communicated to existing clients or customers within 60 days.

In the adopting release, the SEC provides for a review of the effectiveness of Form CRS. In particular, the SEC directs the staff to review a sample of relationship summaries and provide the SEC with results of this review. The directions do not include a description of the scope of the review, the timing of the review, or what the SEC will do with the staff's report once it is provided.

### **Fiduciary Duty Interpretation**

The SEC has issued an interpretive release on the standard of conduct for investment advisers under the Advisers Act. The release brings together in one place the SEC's views on the fiduciary duty that investment advisers owe their clients. These views for the most part are long-standing, although the release does "clarify" certain aspects, as the SEC puts it.

The interpretation explains that an investment adviser's obligation to act in the best interest of its client is an overarching principle that encompasses both the duty of care and the duty of loyalty. This fiduciary duty is made enforceable by the antifraud provisions of the Advisers Act. The fiduciary duty may not be waived, although it will apply in a manner that reflects the agreed-upon scope of the relationship, and the relationship may be shaped by agreement, provided that there is full and fair disclosure and informed consent.

### fiduciarygovernanceblog.com

The SEC clarified that in shaping its agreement with a client, and in considering full and fair disclosure and informed consent, an investment adviser may consider whether the client is institutional or retail. In particular, in describing full and fair disclosure, the SEC recognized that institutional clients "generally have a greater capacity and more resources than retail clients to analyze and understand complex conflicts and their ramifications." Similarly, in describing an adviser's duty of care, the interpretive release distinguishes between institutional and retail clients. In providing advice to institutional clients, the interpretive release recognizes that "the nature and extent of the reasonable inquiry into the client's objectives generally is shaped by the specific investment mandates from those clients."

An adviser's fiduciary duty applies to all investment advice the investment adviser provides to clients, including advice about investment strategy, engaging a sub-adviser, and account type. Similar to Regulation BI and in a clear reference to the fiduciary duty rulemaking efforts of the Department of Labor, the interpretive release clarifies that account type includes "advice about whether to roll over assets from one account (e.g., a retirement account) into a new or existing account that the adviser or an affiliate of the adviser manages."

In the proposal of this interpretation last year, the SEC stated that an adviser must seek to avoid conflicts of interest with its clients. The interpretation as adopted explains that an adviser may satisfy its duty of loyalty by making full and fair disclosure of conflicts of interest and obtaining the client's informed consent. The SEC stated that the requirement to obtain informed consent "does not require advisers to make an affirmative determination that a particular client understood the disclosure and that the client's consent to the conflict of interest was informed. Rather, disclosure should be designed to put a client in a position to be able to understand and provide informed consent to the conflict of interest. A client's informed consent can be either explicit or, depending on the facts and circumstances, implicit." (Emphasis added.) Nonetheless, the interpretation, as adopted, still takes the position that there are some conflicts that cannot be cured through disclosure without providing examples of what such conflicts might be. In particular, the interpretation states that some conflicts "may be of a nature and extent that it would be difficult to provide disclosure to clients that adequately conveys the material facts or the nature, magnitude, and potential effect of the conflict sufficient for a client to consent to or reject it," particularly with regard to retail clients where "it may be difficult to provide disclosure regarding complex or extensive conflicts that is sufficiently specific, but also understandable." In such instances, the interpretation makes clear that disclosure alone is not sufficient, and the adviser should either eliminate the conflict or adequately mitigate it.

Rick Fleming, the SEC's Investor Advocate, released a statement strongly criticizing the interpretive release, arguing that the SEC has taken a step in the wrong direction in its interpretation of the fiduciary duty that investment advisers owe to their clients. In particular, the statement indicates that the final interpretive release "weakens the existing fiduciary standard by suggesting that liability for nearly all conflicts can be avoided through disclosure."<sup>10</sup>

The proposed interpretive release requested comment on several areas of enhanced investment adviser regulation. In particular, the SEC asked for comment on whether to harmonize areas such as licensing and continuing education, provision of account statements, and financial responsibility. The interpretive release indicates that the SEC is continuing to evaluate the comments received in response.

fiduciarygovernanceblog.com

### **Solely Incidental Interpretation**

The SEC also issued an interpretive release on the "Solely Incidental" prong of Section 202(a)(II)(C) of the Advisers Act. That provision excludes from the definition of investment adviser – and thus from the application of the Advisers Act – a broker-dealer "whose performance of such advisory services is solely incidental to the conduct of his business as a broker or dealer and who receives no special compensation" for those services. The SEC did not propose this interpretation when it issued its other proposals last year, but it did request comment on the subject in the release proposing Regulation BI.

The SEC interprets the statutory language to mean that a broker-dealer's provision of advice as to the value and characteristics of securities or as to the advisability of transacting in securities is consistent with the solely incidental prong if the advice is provided in connection with and is reasonably related to the broker-dealer's primary business of effecting securities transactions. If a broker-dealer's primary business is giving advice as to the value and characteristics of securities or the advisability of transacting in securities, or if the advisory services are not offered in connection with or are not reasonably related to the broker-dealer's business of effecting securities transactions, the broker-dealer's advisory services are not solely incidental to its business as a broker-dealer. Whether advisory services provided by a broker-dealer satisfy the solely incidental prong is assessed based on the facts and circumstances surrounding the broker-dealer's business, the specific services offered, and the relationship between the broker-dealer and the customer.

The SEC explains that a broker-dealer's exercise of unlimited discretion (i.e., having responsibility for a customer's trading decisions) would not be solely incidental to the business of a broker-dealer. However, there are situations where a broker-dealer may exercise temporary or limited discretion without being an investment adviser. The SEC also explains that a broker-dealer that agrees to monitor a retail customer's account on a periodic basis for purposes of providing buy, sell, or hold recommendations may still be considered to provide advice in connection with and reasonably related to effecting securities transactions.

### **Compliance Date**

The fiduciary duty and "solely incidental" interpretations will be effective upon publication in the Federal Register. Regulation BI and Form CRS generally will have a compliance date of June 30, 2020.

Even before adoption, press reports indicated that certain constituencies were already considering challenging Regulation BI in court. To the extent that there is a court challenge, it could affect the compliance phase-in timetable.

### **For Further Information**

The foregoing provides our initial, high level summary of the new SEC rulemaking package. We will be following up next week or shortly thereafter with a more detailed analysis.

fiduciarygovernanceblog.com

In addition, various members of Stradley's Fiduciary Governance Group will be hosting a webinar, "Impact of New SEC Regulation Best Interest and Other Standards of Conduct Rules on Broker-Dealers, Investment Advisers and Investment Companies," on July 31 at 1:00 pm Eastern time. Those interested may register at https://www.straffordpub.com/products/impact-of-new-secregulation-best-interest-and-other-standards-of-conduct-rules-on-broker-dealers-investmentadvisers-and-investment-companies-2019-07-31.

In the meantime, feel free to direct any questions you may have about the SEC actions as follows:

#### **Regulation BI**

Larry Stadulis (https://www.stradley.com/professionals/s/stadulis-lawrence-p) Peter Hong (https://www.stradley.com/professionals/h/hong-peter-m) John Baker (https://www.stradley.com/professionals/b/baker-john-m)

### Form CRS

Sara Crovitz (https://www.stradley.com/professionals/c/crovitz-sara) John Baker (https://www.stradley.com/professionals/b/baker-john-m)

### Section 202(a)(11)(C) "solely incidental" Interpretive Release

Larry Stadulis (https://www.stradley.com/professionals/s/stadulis-lawrence-p) Peter Hong (https://www.stradley.com/professionals/h/hong-peter-m)

#### Advisers Act Interpretive Release

Larry Stadulis (https://www.stradley.com/professionals/s/stadulis-lawrence-p) Sara Crovitz (https://www.stradley.com/professionals/c/crovitz-sara) John Baker (https://www.stradley.com/professionals/b/baker-john-m) Alan Goldberg (https://www.stradley.com/professionals/g/goldberg-alan)

# State developments impacting broker-dealer and investment adviser conduct, particularly arising from the SEC rulemaking package

Larry Stadulis (https://www.stradley.com/professionals/s/stadulis-lawrence-p) George Michael Gerstein (https://www.stradley.com/professionals/g/gerstein-george)

### Questions relating to the interrelationship between the SEC rulemaking package and ERISA

George Michael Gerstein (https://www.stradley.com/professionals/g/gerstein-george)

#### Litigation

Bill Mandia (https://www.stradley.com/professionals/m/mandia-william-t)

<sup>&</sup>lt;sup>1</sup> Regulation Best Interest: The Broker-Dealer Standard of Conduct, Release No. 34-86031 at https://www.sec.gov/ rules/final/2019/34-86031.pdf (June 5, 2019).

<sup>&</sup>lt;sup>2</sup> Form CRS Relationship Summary; Amendments to Form ADV, Release Nos. 34-86032, IA-5247 at https://www.sec.gov/rules/final/2019/34-86032.pdf (June 5, 2019); *see also* Appendix B: Form CRS at https://www.sec.gov/rules/final/2019/34-86032-appendix-b.pdf.

### fiduciarygovernanceblog.com

<sup>3</sup> Commission Interpretation Regarding Standard of Conduct of Investment Advisers, Release No. IA-5248 at https://www.sec.gov/rules/interp/2019/ia-5248.pdf (June 5, 2019).

<sup>4</sup> Commission Interpretation Regarding the Solely Incidental Prong of the Broker-Dealer Exclusion from the Definition of Investment Adviser, Release No. IA-5249 at https://www.sec.gov/rules/interp/2019/ia-5249.pdf (June 5, 2019).

<sup>5</sup> See SEC Rulemaking Package Would Impose Best Interest Standard of Conduct at https://www.stradley.com/-/ media/files/publications/2018/04/riskrewardapril202018-pdf.pdf (April 20, 2018).

<sup>6</sup> Jay Clayton, Chairman, SEC, Statement at the Open Meeting on Commission Actions to Enhance and Clarify the Obligations Financial Professionals Owe to our Main Street Investors at https://www.sec.gov/news/public-statement/statement-clayton-060519-iabd (June 5, 2019).

<sup>7</sup> In other words, Regulation BI requires a broker-dealer to disclose the capacity in which the broker-dealer is acting. By using "adviser" or "advisor" in a name or title, the broker-dealer (or associated person) would be falsely disclosing its capacity unless it also is registered as an investment adviser (or is a supervised person of an investment adviser).

<sup>8</sup> "Retail investor" is defined for purposes of Form CRS as "A natural person, or the legal representative of such natural person, who seeks to receive or receives services primarily for personal, family or household purposes." Thus, there is no exception for sophisticated natural person investors.

<sup>9</sup> In particular, broker-dealers that provide recommendations subject to Regulation Best Interest must provide the following disclosure: "When we provide you with a recommendation, we have to act in your best interest and not put our interest ahead of yours." Investment advisers must provide this disclosure: "When we act as your investment adviser, we have to act in your best interest and not put our interest ahead of yours." Dual registrants that prepare a single relationship summary and provide recommendations subject to Regulation Best Interest must disclose the following: "When we provide you with a recommendation as your broker-dealer or act as your investment adviser, we have to act in your best interest and not put our interest ahead of yours."

<sup>10</sup> Rick Fleming, Investor Advocate, SEC, Statement Regarding the SEC's Rulemaking Package for Investment Advisers and Broker-Dealers at https://www.sec.gov/news/public-statement/statement-regarding-secrulemaking-package-investment-advisers-broker-dealers (June 5, 2019).

# A Guide to Regulation Best Interest



# **Fiduciary Governance**

July 1, 2019

# **Risk&Reward**

fiduciarygovernanceblog.com

# A Guide to Regulation Best Interest



The Fiduciary Governance Group has been closely tracking the standard of conduct rulemaking developments out of the Securities and Exchange Commission ("Commission"), Department of Labor ("DOL"), Office of the Comptroller of the Currency, and the states. This new client alert, A Guide to Regulation Best Interest, complements our prior analysis of the SEC rulemaking, particularly our <u>overview</u> of the rulemaking, our analysis of the practical implications of the <u>Investment Adviser Interpretive Release</u>, and our discussion of <u>Form CRS</u>. On July 9 at 2 p.m. (EDT), the Fiduciary Governance Group will host a <u>one-hour webcast</u> where we will examine the SEC rulemaking in its entirety.

On June 5, 2019, the Commission adopted a package of rulemakings and interpretations designed to "enhance and clarify the standards of conduct applicable to broker-dealers and investment advisers, help retail investors better understand and compare the services offered and make an informed choice of the relationship best suited to their needs and circumstances, and foster greater consistency in the level of protections provided by each regime, particularly at the point in time that a recommendation is made."<sup>1</sup> Arguably, the most potentially impactful element of this package is the new Regulation Best Interest, which the Commission adopted as Rule 15*l*-1 under the Securities Exchange Act of 1934 ("Exchange Act").

© 2019 Stradley Ronon Stevens & Young, LLP | www.stradley.com | Philadelphia | Washington | Chicago | New York Information contained in this publication should not be construed as legal advice or opinion or as a substitute for the advice of counsel. The enclosed materials may have been abridged from other sources. They are provided for educational and informational purposes for the use of clients and others who may be interested in the subject matter.

<sup>&</sup>lt;sup>1</sup> SEC Adopts Rules and Interpretations to Enhance Protections and Preserve Choice for Retail Investors in Their Relationships With Financial Professionals, Securities and Exchange Commission (June 5, 2019), <u>https://www.sec.gov/news/press-release/2019-89</u>.

Regulation Best Interest imposes a new federal conduct standard on registered broker-dealers and their associated persons who are natural persons ("Representatives") when making securities transaction or related investment strategy recommendations to their retail customers. Specifically, it requires that a broker-dealer or Representative, when making a covered recommendation, act in the retail customer's best interest and not place its own interests ahead of the customer's interests ("General Conduct Obligation"). To satisfy this General Conduct Obligation, a broker-dealer must satisfy each and every one of the following four core component obligations: (1) provide certain prescribed material disclosures before or at the time of the recommendation about the recommendation and the relationship between the retail customer and the broker-dealer ("Disclosure Obligation"); (2) exercise reasonable diligence, care, and skill in making the recommendation ("Care Obligation"); (3) establish, maintain, and enforce policies and procedures reasonably designed to address conflicts of interest ("Conflict of Interest Obligation"); and (4) establish, maintain, and enforce policies and procedures reasonably designed to achieve compliance with Regulation Best Interest ("Compliance Obligation") (collectively, "Component Obligations"). Thus, a failure to satisfy any single one of the Component Obligations constitutes a violation of Regulation Best Interest.

In adopting Regulation Best Interest, the Commission ultimately "declined to subject broker-dealers to a wholesale and complete application of the existing fiduciary standard under the Adviser Act because it is not appropriately tailored to the structure and characteristics of the broker-dealer business model (*i.e.*, transaction-specific recommendations and compensation), and would not properly take into account, and build upon, existing obligations that apply to broker-dealers . . . .<sup>22</sup> Moreover, in the Commission's view, "this approach would significantly reduce retail investor access to differing types of investment services and products, reduce retail investor choice in how to pay for those products and services, and increase costs for retail investors of obtaining investment recommendations."<sup>3</sup> Instead, the enhancements contained in Regulation Best Interest "are designed to improve investor protection by enhancing the quality of broker-dealer recommendations to retail customers and reducing the potential harm to retail customers that may be caused by conflicts of interest."<sup>4</sup> Together with the other elements of the rulemaking and interpretive package, Regulation Best Interest also was designed to help retail customers better understand and compare the services offered by broker-dealers and investment advisers and make an informed choice of the relationship best suited to their needs and circumstances, provide clarity with respect to the standards of conduct applicable to investment advisers and broker-dealers, and foster greater consistency in the level of protections provided by each regime, particularly at the point in time that a recommendation is made.

This Client Alert provides an overview of Regulation Best Interest. The Alert is divided into five sections. Section I discusses Regulation Best Interest generally. Section II discusses at some length each of the four Component Obligations that must be satisfied to demonstrate compliance with Regulation Best Interest. Section III examines a number of other important collateral issues under and related to Regulation Best Interest. Section IV discusses required records and recordkeeping related to Regulation Best Interest. Finally, Section V provides a suggested framework for ensuring that registered broker-dealers are fully compliant with Regulation Best Interest on or before the fast-approaching compliance date of June 30, 2020.

### <u>Key Takeaways</u>

• Regulation Best Interest requires that a broker-dealer or its Representative, when making a recommendation, act in the retail customer's best interest and not place its own interests ahead of the customer's interests, which is satisfied only if the broker-dealer or Representative complies with the four specified component obligations, referred to as the Disclosure Obligation, the Care Obligation, the Conflict of Interest Obligation, and the Compliance Obligation.

<sup>3</sup> Id.

<sup>&</sup>lt;sup>2</sup> Regulation Best Interest: The Broker-Dealer Standard of Conduct, Exchange Act Rel. No. 34-86031 (June 5, 2019), https://www.sec.gov/rules/final/2019/34-86031.pdf ("Adopting Release").

<sup>&</sup>lt;sup>4</sup> *Id*. at 16.

### fiduciarygovernanceblog.com

- Regulation Best Interest applies to broker-dealers and Representatives that make recommendations of
  securities transactions and investment strategies to retail customers. Recommendations include
  recommendations of account types and rollovers or transfers of assets (*e.g.*, to roll over or transfer assets in a
  workplace retirement plan to an individual retirement account). Recommendations also include implicit hold
  recommendations resulting from agreed-upon account monitoring.
- A retail customer is a natural person (or legal representative) who receives a recommendation and uses it primarily for personal, family or household purposes. There is no carve-out for wealthy individuals, so Warren Buffett is a retail customer, assuming his brokerage services are not for commercial or business purposes.
- With limited exceptions (*e.g.*, for commodity trading advisors), a broker-dealer or Representative cannot use the title "adviser" or "advisor" unless that person is also an investment adviser or a supervised person of an investment adviser. The rationale is that the title would be inconsistent with the disclosure of the capacity in which the person acts.
- Before or at the time of the recommendation, a broker-dealer or Representative must disclose, in writing, all material facts about the scope and terms of its relationship with the customer and all material facts relating to conflicts of interest that are associated with the recommendation.
- A broker-dealer or Representative must exercise reasonable diligence, care, and skill when making a
  recommendation to a retail customer. The broker-dealer or Representative must understand potential risks,
  rewards, and costs associated with the recommendation. While costs must always be considered in making a
  recommendation, they should be considered in light of other factors and the retail customer's investment
  profile; there is no requirement to recommend only the lowest cost option.
- When recommending a series of transactions, the broker-dealer or Representative must have a reasonable basis to believe that the transactions taken together are not excessive, even if each is in the customer's best interest when viewed in isolation.
- A broker-dealer must establish, maintain, and enforce reasonably designed written policies and procedures addressing conflicts of interest associated with its recommendations to retail customers. These policies and procedures must be reasonably designed to identify all such conflicts and at a minimum disclose or eliminate them.
- The policies and procedures must be reasonably designed to mitigate conflicts of interests that create an incentive for an associated person to place his or her interests or the interest of the firm ahead of the retail customer's interest.
- Where a broker-dealer places material limitations on recommendations (*e.g.*, offering only proprietary or other limited range of products), the policies and procedures must be reasonably designed to disclose the limitations and associated conflicts and to prevent the limitations from causing the associated person or broker-dealer to place the person's or the firm's interests ahead of the customer's interest.
- Sales contests, sales quotas, bonuses, and non-cash compensation are prohibited if they are based on (1) the sale of specific securities or specific types of securities (2) within a limited period of time. Other incentives and practices that are not explicitly prohibited are permitted, provided that the broker-dealer establishes reasonably designed policies and procedures to disclose and mitigate the incentive created to the Representative, and the Care and Disclosure Obligations are complied with.
- A broker-dealer must establish, maintain, and enforce written policies and procedures reasonably designed to achieve compliance with Regulation Best Interest as a whole. Thus, a broker-dealer's policies and procedures must address not only conflicts of interest but also compliance with its Disclosure and Care Obligations.

 Compliance with Regulation Best Interest will not alter a broker-dealer's obligations under the general antifraud provisions of the federal securities laws. Regulation Best Interest applies in addition to any applicable securities laws and regulations.

#### I. General Obligation

Regulation Best Interest requires that a broker-dealer or Representative, when making a recommendation, act in the retail customer's best interest and not place the financial or other interest of the person making the recommendation ahead of the customer's interests. Despite requiring broker-dealers to act in the "best interest" of a retail customer, the Commission chose not to expressly define "best interest." Instead, the Commission has stated that whether a broker-dealer has acted in the best interest of a retail customer will turn on an objective assessment of the facts and circumstances of how the specific components of Regulation Best Interest, including the Component Obligations, are satisfied at the time that the recommendation is made (and not in hindsight).

In crafting Regulation Best Interest, the Commission drew upon key principles underlying fiduciary obligations, including those that apply to investment advisers under the Investment Advisers Act of 1940 ("Advisers Act"), but recognized that broker-dealers and investment advisers have different types of relationships with investors, offer different services, and have different compensation models when providing investment recommendations or investment advisory services to customers. Specifically, the Commission recognized that a broker-dealer's recommendations may include recommending transactions where the broker-dealer is buying securities from or selling securities to retail customers on a principal basis or recommending proprietary products, while investment advisers typically provide ongoing, regular advice and services in the context of a broad investment portfolio management and are compensated based on the value of assets under management, a fixed fee, or other arrangement. In adopting Regulation Best Interest, the Commission sought to maintain the difference in offerings that broker-dealers and investment advisers provide so as to offer retail customers with choices regarding the types of relationships they can have, the services they can receive, and how they can pay for those services.

### A. Natural Person Who Is an Associated Person

As noted above, the General Obligation applies to recommendations made by a broker-dealer or a "natural person who is an associated person" of a broker-dealer. Section 3(a)(18) of the Exchange Act generally defines "associated person of a broker or dealer" as "any partner, officer, or director, or branch manager of such broker or dealer (or any person occupying a similar status or performing similar functions); any person directly or indirectly controlling, controlled by, or under common control with such broker or dealer; or any employee of such broker or dealer." In limiting the term to only a "natural person" who is an "associated person," the Commission sought to exclude affiliated entities of the broker-dealer that are not themselves broker-dealers, as they are not the intended focus of Regulation Best Interest.

### B. Recommendation of Any Securities Transaction or Investment Strategy Involving Securities

#### 1. Recommendation

The imposition of the General Obligation hinges upon whether a "recommendation" has been made of "any securities transaction or investment strategy involving securities."<sup>5</sup> However, despite some commenters' request for a formal definition of "recommendation" that sets out what is and what is not a "recommendation," the Commission determined not to formally define the term but chose instead, consistent with the Proposing Release, to rely on the existing regulatory framework that would provide clarity to broker-dealers and maintain efficiencies for broker-dealers with established infrastructures that already rely on the term. Therefore, a determination of whether a broker-dealer has made a "recommendation" that triggers application of Regulation Best Interest turns on the facts and circumstances of the particular situation. The Commission identified two factors that should be

<sup>&</sup>lt;sup>5</sup> Adopting Release, *supra* note 2, at 78.

### fiduciarygovernanceblog.com

considered, among other facts and circumstances, when determining whether a recommendation has taken place: (1) whether a communication reasonably could be viewed as a "call to action"; and (2) whether a communication "reasonably would influence an investor to trade a particular security or group of securities."<sup>6</sup> In addition, the more individually tailored a communication is to a specific customer or a targeted group of customers about a security or group of securities, the greater the likelihood of the communication being viewed as a "recommendation." However, the Commission generally views the following types of communications as not being recommendations of any securities transaction or investment strategy involving securities:

- Certain general financial and investment information, including:
  - basic investment concepts, such as risk and return, diversification, dollar cost averaging, compounded return, and tax deferred investment;
  - historic differences in the return of asset classes (*e.g.*, equities, bonds, or cash) based on standard market indices;
  - effects of inflation;
  - o estimates of future retirement income needs; and
  - o assessment of a customer's investment profile.
- Descriptive information about an employer-sponsored retirement or benefit plan, participation in the plan, the benefits of plan participation, and the investment options available under the plan.
- Asset allocation models that are:
  - o based on generally accepted investment theory;
  - accompanied by disclosures of all material facts and assumptions that may affect a reasonable investor's assessment of the asset allocation model or any report generated by such model;
  - in compliance with FINRA Rule 2214 (Requirements for the Use of Investment Analysis Tools) if the asset allocation model is an "investment analysis tool" covered by FINRA Rule 2214.
- Interactive investment materials that incorporate the above.

#### 2. Any Securities Transaction or Investment Strategy Involving Securities

Regulation Best Interest applies to a recommendation of "any securities transaction or investment strategy involving securities (including account recommendations) to a retail customer." The inclusion of "account recommendations" was one of the key enhancements adopted by the Commission.<sup>7</sup> The Commission interprets "account recommendations" to include recommendations of securities account types generally (*e.g.*, to open an IRA or other brokerage account), as well as recommendations to roll over or transfer assets from one type of account to another (*e.g.*, a workplace retirement plan account to an IRA). Such account recommendations are "investment strategies involving securities" for purposes of Regulation Best Interest, regardless of whether they are tied to a specific securities transaction. The Commission believes that because customers may rely on firms' and Representatives' investment expertise and knowledge the broker-dealer should be responsible for such recommendations, regardless of whether those recommendations result in transactions or generate transaction-based compensation.

<sup>&</sup>lt;sup>6</sup> *Id.* at 79-80.

<sup>&</sup>lt;sup>7</sup> Id. at 85.

### fiduciarygovernanceblog.com

Consistent with existing broker-dealer regulation, Regulation Best Interest applies to explicit recommendations to hold a security or securities or to stay in a particular account. The Commission also stated its view that "any securities transaction or investment strategy involving securities" not only includes explicit hold recommendations, but also implicit hold recommendations that are the result of an explicit agreement between the broker-dealer and retail customer. While not required to do so, broker-dealers may enter into an agreement with a retail customer to monitor that customer's account. Such an agreement would constitute "recommendations of any securities transaction or investment strategy" because there is an implicit recommendation not to buy, sell, or exchange assets, *i.e.*, to hold, during the monitoring process; and that hold recommendation is also viewed as a recommendation of an investment strategy. The Commission's position differs from existing FINRA guidance, which generally provides that the FINRA suitability rule does not cover an implicit hold recommendation.<sup>8</sup> The Commission explained that "implicit" hold recommendations in the context of an agreement to monitor a customer's account are similar to explicit hold recommendations that are considered "investment strategies" because they would constitute the type of recommendations that retail customers would likely be deem to be a "call to action." tantamount to an explicit recommendation to hold.<sup>9</sup> Any voluntary review by a broker-dealer in the absence of an express agreement, however, would not be subject to Regulation Best Interest.

#### C. Retail Customer

Regulation Best Interest defines "retail customer" to mean, a natural person, or the legal representative of such natural person, who receives a recommendation and who uses the recommendation primarily for personal, family, or household purposes. The term "legal representative" means a non-professional legal representative of a natural person. The Commission noted that this change and clarification provides more certainty that institutions and certain professional fiduciaries are not covered for purposes of Regulation Best Interest. It retains, however, coverage of certain legal entities (*e.g.*, trusts that represent the assets of a natural person) specifically identified in the Proposing Release as "retail customers" within the scope of Regulation Best Interest. The term "retail customer" does not exclude certain high-net-worth natural persons, as was suggested by some commenters to match the current FINRA exclusion of such natural persons from customer-specific suitability requirements.

The Commission views a "legal representative" of a natural person to only cover non-professional legal representatives (*e.g.*, a non-professional trustee that represents the assets of a natural person and similar representatives such as executors, conservators, and persons holding a power of attorney for a natural person), thereby excluding certain institutions from Regulation Best Interest's coverage. In capturing non-professional legal representatives within the definition of retail customer, the Commission provided the protections of Regulation Best Interest to non-professional persons who are acting on behalf of natural persons but who are not regulated financial services industry professionals retained by natural persons and their legal representatives who rely directly on the broker-dealer for the recommendation. The professionals so excluded include registered investment advisers and broker-dealers, corporate fiduciaries (*e.g.*, banks, trust companies and similar financial institutions) and insurance companies, and the employees or other regulated representatives of such advisers, broker-dealers, corporate fiduciaries and insurance companies.<sup>10</sup>

<sup>&</sup>lt;sup>8</sup> See FINRA Rule 2111 (Suitability) FAQ, <u>https://www.finra.org/industry/faq-finra-rule-2111-suitability-faq</u>.

<sup>&</sup>lt;sup>9</sup> The Commission reiterated that Regulation Best Interest does not change the scope of account monitoring that broker-dealers may agree to provide, nor does it change the scope of activities that would come within the "solely incidental" prong of the broker-dealer exclusion to the definition of "investment adviser" in the Advisers Act. Adopting Release, *supra* note 2, at 106.

<sup>&</sup>lt;sup>10</sup> *Id.* at 113-14.

### fiduciarygovernanceblog.com

The Commission also noted that it interprets "personal, family or household purposes" to mean that *any* recommendation to a natural person for his or her account would be subject to Regulation Best Interest, other than recommendations to natural persons seeking these services for commercial or business purposes. Accordingly, under this interpretation, "personal, family or household purposes" would not include, for example, an employee seeking services for an employer or an individual who is seeking services for a small business or on behalf of another non-natural person entity such as a charitable trust.

The Commission also confirmed that "personal, family or household purposes" would cover retirement accounts, as retirement savings is a personal, household, or family purpose. It would not apply where the recommendation is not made to an individual participant. Accordingly, the definition of retail customer will include a natural person receiving recommendations for his or her own retirement account, including but not limited to IRAs and individual accounts in workplace retirement plans, such as 401(k) plans and other tax-favored retirement plans. Additionally, the Commission stated that it does not believe that workplace retirement plans or their representatives and service providers generally fall within the definition of retail customer for purposes of Regulation Best Interest because the workplace retirement plan is not a natural person. Therefore, a workplace retirement plan representative is not a non-professional representative of a natural person that is receiving a recommendation directly from a broker-dealer for "personal, family, or household purposes." To the extent that a plan representative who decides service arrangements for a workplace retirement plan is a sole proprietor or other self-employed individual who will participate in the plan, the plan representative would be a retail customer for purposes of Regulation Best Interest because for purposes of Regulation Best Interest because the workplace retirement plan is a sole proprietor or other self-employed individual who will participate in the plan, the plan representative would be a retail customer for purposes of Regulation Best Interest to the extent the sole proprietor or self-employed individual receives recommendations directly from a broker-dealer primarily for personal, family, or household purposes.

Regulation Best Interest applies to a retail customer that both receives a recommendation of any securities transaction or investment strategy involving securities by a broker-dealer and that uses that recommendation primarily for personal, family, or household purposes, and not simply those recommendations for which a broker-dealer receives compensation. Accordingly, "a retail customer 'uses' a recommendation of a securities transaction or investment strategy involving securities when, as a result of the recommendation (1) the retail customer opens a brokerage account with the broker-dealer, regardless of whether the broker-dealer receives compensation; (2) the retail customer has an existing account with the broker-dealer and receives a recommendation from the broker-dealer, regardless of whether the broker-dealer receives or will receive compensation, directly or indirectly, as a result of that recommendation; or (3) the broker-dealer receives or will receive compensation, directly or indirectly as a result of that recommendation, even if that retail customer does not have an account at the firm."<sup>11</sup> When a retail customer opens or has an existing account with a broker-dealer the retail customer has a relationship with the broker-dealer and is therefore in a position to "use" (*i.e.*, accept or reject) the brokerdealer's recommendation. In this context, tying "use" solely to a broker-dealer's receipt of compensation would inappropriately result in Regulation Best Interest not applying to the broker-dealer's recommendations to hold securities positions or to maintain an investment strategy (such as account type), recommendations to open an account, or recommendations that may ultimately be rejected by the retail customer.

Whether the recommendation complies with Regulation Best Interest will be evaluated based on the circumstances that existed at the time the recommendation was made to the retail customer. Accordingly, broker-dealers should carefully consider the extent to which Representatives can make recommendations to prospective retail customers (*i.e.*, that have received, but not yet "used" the recommendation as noted above) in compliance with Regulation Best Interest, including having gathered sufficient information that would enable them to comply with Regulation Best Interest at the time the recommendation is made, should the prospective retail customer use the recommendation.<sup>12</sup>

<sup>&</sup>lt;sup>11</sup> *Id.* at 120-21.

<sup>&</sup>lt;sup>12</sup> Based on comments received in response to the Proposing Release, the definitions in Form CRS and Regulation Best Interest have been revised to generally conform to each other, consistent with the Commission's respective goals in each of these rulemakings. The definition of "retail customer" for purposes of Regulation Best Interest has been revised to apply only to natural persons, not all persons, in line with the definition of "retail investor" for purposes of Form CRS. In addition, the definition in Form CRS as adopted now includes the "personal, family or household purposes" qualifier. The Commission clarified that the definition of "retail customer" does not apply to prospective customers who do not receive and use recommendations from a broker-dealer. This distinction reflects differences between the point in time the Relationship

#### D. Treatment of Dual-Registrants

Regulation Best Interest applies to a dual-registrant and its Representatives only when making recommendations in a broker-dealer capacity.<sup>13</sup> Thus, it does not apply, for example, where an investment adviser representative of a dual-registrant provides investment advice to a client in his or her investment advisory capacity. This is so, even if the retail customer has a brokerage relationship with the dual-registrant or the dual-registrant executes the transaction in its brokerage capacity. Similarly, as proposed, the Commission confirmed that a dual-registrant is an investment adviser solely with respect to those accounts for which a dual-registrant provides investment advice or receives compensation that subjects it to the Advisers Act.

Determining the capacity in which a dual-registrant is making a recommendation is a facts and circumstances test, with no one factor being determinative. However, the Commission identified as factors the type of account, how the account is described, the type of compensation and the extent to which the dual-registrant made clear to the customer or client the capacity in which it was acting.

Regulation Best Interest would not apply when a dually registered financial professional of a dually registered broker-dealer and investment adviser, who is acting in the capacity of an investment adviser, recommends a fee-based account. However, the dually registered financial professional would need to comply with the Advisers Act as well as the requirements with respect to Form CRS for the firm. Additionally, the Commission stated that Regulation Best Interest would apply if the financial professional in his or her brokerage capacity (disclosed pursuant to the Disclosure Obligation), provides a recommendation of a securities transaction or investment strategy involving securities to the family in the course of the holistic review.

#### II. Component Obligations

As adopted, to comply with Regulation Best Interest, each of the four specified Component Obligations must be met. According to the Commission, whether a broker-dealer has acted in the retail customer's best interest under the General Obligation turns on an objective assessment of the facts and circumstances of how these specific Component Obligations are satisfied at the time the recommendation is made.<sup>14</sup> Each of the four Obligation are discussed below.

#### A. Disclosure Obligation

The Commission determined it was necessary to impose a more explicit and broader disclosure obligation on broker-dealers than that which currently exists under the federal securities laws and SRO rules. To that end, Regulation Best Interest requires a broker-dealer, or a Representative, prior to or at the time of the recommendation, to provide to a retail customer, in writing, "full and fair disclosure of all material facts related to the scope and terms of the relationship" with the retail customer and "all material facts relating to conflicts of interest associated with the recommendation."<sup>15</sup> A Representative of a broker-dealer may generally rely on the disclosure of their broker-dealer, unless they know or should have known that the broker-dealer's disclosure is insufficient. For example, "if a [Representative] of a broker-dealer that offers a full range of securities products is licensed solely as a Series 6 Registered Representative, and can sell only mutual funds, variable annuities and other enumerated products, that limitation on the scope of services provided by the particular [Representative] must be sufficiently clear in the broker-dealer's

Summary is delivered to an investor and when the obligations of broker-dealers pursuant to Regulation Best Interest attach. Also of note, fund underwriters are not required to deliver Form CRS so long as they do not make recommendations.

<sup>&</sup>lt;sup>13</sup> Adopting Release, *supra* note 2, at 126.

<sup>&</sup>lt;sup>14</sup> Adopting Release at 35-36.

<sup>&</sup>lt;sup>15</sup> *Id.* at 37.

#### fiduciarygovernanceblog.com

disclosures; otherwise additional clarifying disclosure by the associated person would be necessary."<sup>16</sup> The standard for materiality for purposes of the Disclosure Obligation is consistent with the familiar standard articulated in *Basic v. Levinson*,<sup>17</sup> where a fact is material if there is "a substantial likelihood that a reasonable shareholder would consider it important."<sup>18</sup> In addition to the new Disclosure Obligation, the Commission now requires broker-dealers and investment advisers to deliver to retail investors a Relationship Summary, the purpose of which is to provide an initial layer of disclosure.<sup>19</sup>

#### 1. Material Facts Regarding Scope and Terms of the Relationship

The Commission identified as a material fact the capacity in which a broker-dealer is acting, which must be affirmatively disclosed as it relates to the scope and terms of the client relationship.<sup>20</sup> Disclosure must include whether the broker-dealer or such natural person is acting as a broker-dealer or a Representative of a broker-dealer with respect to the recommendation. Representatives of a dually registered firm who are not dually licensed but only offer broker-dealer services through the firm must be especially careful to disclose that they are acting in their capacity as a Representative of a broker-dealer not also registered as an investment adviser or by a financial professional that is not also a supervised person of an investment adviser, generally will be considered to be a violation of the Disclosure Obligation under Regulation Best Interest.

Regulation Best Interest explicitly requires the disclosure of all material fees and costs. The Disclosure Obligation does not mandate individualized fee disclosure specific to each retail customer; instead, broker-dealers should disclose facts about material fees and costs in standardized numerical terms and narrative disclosures, "such as standardized or hypothetical amounts, dollar or percentage ranges, and explanatory text" in order to make clear why a fee is being imposed and when the fee is to be charged.

The type and scope of services a broker-dealer provides to its retail customers are also considered by the Commission to be "material facts relating to the scope and terms of the relationship." Regulation Best Interest also explicitly requires that a broker-dealer include "any material limitations on the securities or investment strategies involving securities that may be recommended to the retail customer." Material limitations placed on the securities or investment strategies involving securities could include, for example, "recommend[ing] only proprietary products, recommend[ing] only products from a select group of issuers, or mak[ing] IPOs available only to certain clients."<sup>21</sup> The Commission also considers as material the facts, account monitoring services, account balance requirements, and risks associated with the recommendation.

#### 2. Material Facts Regarding Conflicts of Interest

The Commission defined conflicts of interest that are associated with recommendations as "an interest that might incline a broker, dealer, or a natural person who is an associated person of a broker or dealer – consciously or unconsciously – to make a recommendation that is not disinterested."<sup>22</sup> The Disclosure Obligation requires the disclosure of "material facts" regarding such

<sup>18</sup> *Id*.

<sup>&</sup>lt;sup>16</sup> *Id.* at 141-42.

<sup>&</sup>lt;sup>17</sup> Basic, Inc. v. Levinson, 485 U.S. 224 (1988).

<sup>&</sup>lt;sup>19</sup> Form CRS Relationship Summary; Amendments to Form ADV, Exchange Act Rel. Nos. 34-86032 (June 5, 2019), https://www.sec.gov/rules/final/2019/34-86032.pdf.

<sup>&</sup>lt;sup>20</sup> Adopting Release, *supra* note 2, at 178.

<sup>&</sup>lt;sup>21</sup> *Id*. at 179.

<sup>&</sup>lt;sup>22</sup> *Id.* at 36.

### fiduciarygovernanceblog.com

conflicts of interest. Without giving concrete examples of what they would view as "material facts," the Commission stated that the conflicts of interest identified in the Relationship Summary provide a useful starting point for the identification of material facts that need to be disclosed. In particular, the Commission views how a broker-dealer's investment professionals are compensated, and the conflicts associated with those arrangements, as material facts relating to conflicts of interest.

Regulation Best Interest allows broker-dealers to meet the Disclosure Obligation by making certain required disclosures of information regarding conflicts of interest to their customers at the beginning of a relationship, and this form of disclosure may be standardized. However, any subsequent recommendations must be made with additional disclosure. For example, "with regard to mutual fund transactions and holdings, a broker-dealer might disclose broadly that it is compensated by funds out of product fees or by the funds' sponsors, and that such compensation gives it an incentive to recommend certain products over other products for which the broker-dealer receives less compensation."<sup>23</sup> Later, when a broker-dealer recommends a particular fund, "it could provide more specific detail about compensation arrangements, for example revenue sharing associated with the fund family."<sup>24</sup>

#### 3. Full and Fair Disclosure

The Commission adopted as part of the Disclosure Obligation the requirement of full and fair disclosure of all material facts relating to the scope and terms of a relationship with a retail customer and all material facts relating to conflicts of interest associated with a recommendation.<sup>25</sup> This requirement of broker-dealers is similar to the disclosure standard applicable to investment advisers under the Advisers Act. The Commission decided not to require any standard written disclosures; instead, disclosure can be made to accommodate the structure and characteristics of a particular broker-dealer. The Commission also clarified that broker-dealers' compliance with the Disclosure Obligation will be measured against a negligence standard, not against a standard of strict liability, consistent with the Proposing Release.

### 4. Oral Disclosure or Disclosure After a Recommendation

Regulation Best Interest requires that the Disclosure Obligation be made "in writing." However, in recognition of the challenges associated with providing written disclosure in each instance that may be required, the Commission clarified that a broker-dealer could orally clarify the capacity in which it is acting at the time of the recommendation if it had already provided written disclosure regarding its capacity as well as the method it planned to use to clarify its capacity at the time of the recommendation. Similarly, as set forth in the Proposing Release, the Commission stated that where a broker-dealer determines to provide an initial, more general disclosure followed by specific information in a subsequent disclosure that is provided after the recommendation, the initial disclosure should address when and how the broker-dealer would provide more specific information regarding the fact or conflict in a subsequent disclosure. The Commission also stated that a brokerdealer may satisfy the Disclosure Obligation by making a supplemental oral disclosure not later than the time of the recommendation, provided that the broker-dealer maintains a record of the fact that oral disclosure was provided to the retail customer. In addition, where existing regulations permit disclosure after a recommendation is made (e.g., trade confirmation, prospective delivery), a brokerdealer may satisfy its Disclosure Obligation by providing such document to the retail customer after the recommendation is made. Again, broker-dealers must provide an initial disclosure in writing that identifies the material facts and describes the process through which such fact may be supplemented, clarified, or updated.

24

<sup>&</sup>lt;sup>23</sup> *Id.* at 204.

<sup>&</sup>lt;sup>24</sup> *Id*.

<sup>&</sup>lt;sup>25</sup> *Id.* at 205.

#### B. Care Obligation

The Care Obligation incorporates and enhances existing suitability requirements applicable to brokerdealers under the federal securities laws by, among other things, imposing a "best interest" requirement that will require a broker-dealer to not place its own interest ahead of the retail customer's interest when making recommendations. As adopted, the Care Obligation requires that a broker-dealer or a Representative, in making a recommendation, exercise reasonable diligence, care, and skill to (1) understand potential risks, rewards, and costs associated with the recommendation, and have a reasonable basis to believe that the recommendation could be in the best interest of at least some retail customers; (2) have a reasonable basis to believe that the recommendation is in the best interest of a particular retail customer based on that retail customer's investment profile and the potential risks, rewards, and costs associated with the recommendation and does not place the financial or other interest of the broker-dealer or Representative ahead of the interest of the retail customer; and (3) have a reasonable basis to believe that a series of recommended transactions, even if in the retail customer's best interest when viewed in isolation, is not excessive and is in the retail customer's best interest when taken together in light of the retail customer's investment profile and does not place the financial or other interest of the broker-dealer or Representative making the series of recommendations ahead of the interest of the retail customer.26

In response to comments, the Commission altered multiple aspects of the Care Obligation as proposed. First, the "prudence" requirement has been deleted as the Commission viewed "prudence" as duplicative and confusing when included with "reasonable diligence, care, and skill." Second, the Commission added a requirement that the broker-dealer exercise reasonable diligence, care, and skill to understand potential costs associated with the recommendation. The Commission noted that the standard does not necessarily require the lowest cost option, and that while cost is an important factor that always needs to be taken into consideration in making a recommendation, it is not the only factor. Finally, the Commission added that broker-dealers must have a reasonable basis to believe that the recommendation "does not place the financial or other interest of the broker-dealer ahead of the interest of the retail customer." This addition is intended to make clear that while a broker-dealer typically will have some interest in a recommendation, the broker-dealer cannot put that interest ahead of the retail customer's interest when making the recommendation.

#### 1. Reasonable Diligence, Care, and Skill

What would constitute reasonable diligence, care, and skill under the Care Obligation varies depending on, among other things, the complexity of and risks associated with the recommended security or investment strategy and the broker-dealer's familiarity with the recommended security or investment strategy.<sup>27</sup> While every inquiry will be specific to the particular broker-dealer and the recommended security or investment strategy, broker-dealers generally should consider important factors such as the security's or investment strategy's investment objectives, characteristics (including any special or unusual features), liquidity, volatility, and likely performance in a variety of market and economic conditions; and the expected return of the security or investment strategy; as well as any financial incentives to recommend the security or investment strategy. Together, this inquiry allows the broker-dealer to develop a sufficient understanding of the security or investment strategy and to be able to reasonably believe that it could be in the best interest of at least some retail customers.

<sup>&</sup>lt;sup>26</sup> *Id*. at 15.

<sup>&</sup>lt;sup>27</sup> *Id.* at 262.

While the Care Obligation does not require broker-dealers to document the basis for a recommendation, broker-dealers may choose to take a risk-based approach when deciding whether or not to document certain recommendations. Similarly, broker-dealers may consider using existing compliance measures, such as generating and reviewing exception reports that identify transactions that fall outside of firm-specified parameters to help evaluate and review for compliance with the Care Obligation.

The Care Obligation also requires a broker-dealer to exercise reasonable diligence, care, and skill to have a reasonable basis to believe that a series of recommended transactions, even if in the retail customer's best interest when viewed in isolation, is not excessive and is in the retail customer's best interest when taken together in light of the retail customer's investment profile. This requirement incorporates and enhances a broker-dealer's existing "quantitative suitability" obligation by applying the requirement irrespective of whether a broker-dealer exercises actual or *de facto* control over a customer's account, thereby making the obligation consistent with the current requirements for "reasonable basis suitability" and "customer specific suitability."

The Commission noted that the fact that a customer may have some knowledge of financial markets or some "control" should not absolve the broker-dealer of the ultimate responsibility to have a reasonable basis to believe that any recommendations it makes are in the best interest of the retail customer. Where a retail customer expresses a desire for "active trading," a broker-dealer may take this factor into consideration when evaluating a recommendation; however, the broker-dealer will nevertheless need to reasonably believe that a series of recommended transactions is in the best interest of the retail customer. Regulation Best Interest does not require a broker-dealer to refuse to accept a customer's order that is contrary to the broker-dealer's recommendation. Nor does Regulation Best Interest apply to self-directed or otherwise unsolicited transactions by a retail customer, whether or not he or she also receives separate recommendations from the broker-dealer.

Consistent with the other components of the Care Obligation, when a series of transactions is recommended to a retail customer, a broker-dealer must evaluate whether the series of recommended transactions places the broker-dealer's interest ahead of the retail customer's – this is true for both the Representative making the recommendation, as well as for the firm. This will necessarily depend on the facts and circumstances of each particular recommendation, and of each particular series of transactions; however, the Commission noted that, as part of developing a retail customer's investment profile, a broker-dealer must exercise reasonable diligence to ascertain the retail customer's other investments.

Finally, whether recommendations constitute a "series" of recommended transactions depends on the facts and circumstances, and needs to be evaluated with respect to a particular retail customer. In other words, a broker-dealer would need to reasonably believe that the level of trading (series of recommended transactions) is appropriate for a particular retail customer, and thus a bright line definition across all retail customers would be unworkable. Moreover, providing a bright line definition could encourage firms to focus on a particular number of transactions rather than focusing on ensuring that a series of recommended transactions, taken together, are in the best interest of the retail customer. Finally, a "series" of recommended transactions is an established term under the federal securities laws and SRO rules that is evaluated in concert with existing guideposts, such as turnover rate, cost-to-equity ratio, and use of in-and-out trading, which have been developed over time and which serve as indicators of excessive trading.

#### 2. Retail Customer Investment Profile

Regulation Best Interest requires a "Retail Customer Investment Profile" to include, but not be limited to, "the retail customer's age, other investments, financial situation and needs, tax status, investment objectives, investment experience, investment time horizon, liquidity needs, risk tolerance, and any other information the retail customer may disclose to the broker, dealer, or a natural person who is an

associated person of a broker or dealer in connection with a recommendation."<sup>28</sup> Broker-dealers are required to exercise "reasonable diligence" to ascertain the retail customer's investment profile in satisfying the Care Obligation, and when retail customer information is unavailable despite a broker-dealer's reasonable diligence to obtain such information, a broker-dealer should consider whether it has sufficient understanding of the retail customer to properly evaluate whether the recommendation is in the retail customer's best interest. Furthermore, a broker-dealer would not meet its Care Obligation if it made a recommendation to a retail customer for whom it lacks sufficient information to have a reasonable basis to believe that the recommendation is in the best interest of that retail customer's investment profile.

Broker-dealers must obtain and analyze enough customer information to have a reasonable basis to believe that the recommendation is in the best interest of the particular retail customer. The significance of specific types of customer information generally depends on the facts and circumstances of the particular case, including the nature and characteristics of the product or strategy at issue. Where retail customer information is unavailable despite a broker-dealer's reasonable diligence, the broker-dealer should carefully consider whether it has a sufficient understanding of the retail customer to properly evaluate whether the recommendation is in the best interest of that retail customer. In addition, a broker-dealer generally should make a reasonable effort to ascertain information regarding an existing customer's investment profile prior to the making of a recommendation on an "as needed" basis - that is, where a broker-dealer knows or has reason to believe that the customer's investment profile has changed. The reasonableness of a broker-dealer's efforts to collect information regarding a customer's investment profile information depends on the facts and circumstances of a given situation, and the importance of each factor may vary depending on the facts and circumstances of the particular case. Under Regulation Best Interest, as with the approach under FINRA's suitability rule, broker-dealers may generally rely on a retail customer's responses absent "red flags" indicating that the information is inaccurate.

### 3. Reasonably Available Alternatives and Otherwise Identical Securities

The Commission noted that as part of determining whether a broker-dealer has a reasonable basis to believe that a recommendation is in the best interest of the retail customer, a broker-dealer generally should consider reasonably available alternatives offered by the broker-dealer.<sup>29</sup> In terms of conducting such an evaluation, a broker-dealer does not have to conduct an evaluation of every possible alternative, either offered outside of the firm (such as where the firm offers only proprietary or other limited range of products) or available on the firm's platform. A broker-dealer will not be required to recommend the single "best" of all possible alternatives that might exist, in part because many different options may in fact be in the retail customer's best interest.

In particular, the Commission does not require a Representative to be familiar with every product on a broker-dealer's platform, particularly where a broker-dealer operates in an open architecture framework or otherwise operates a platform with a large number of products or options. Consistent with the Compliance Obligation, a broker-dealer should have a reasonable process for establishing and understanding the scope of such "reasonably available alternatives" that would be considered by particular Representatives or groups of Representatives (*e.g.*, groups that specialize in particular product lines) in fulfilling the reasonable diligence, care, and skill requirements under the Care Obligation.

What will be a reasonable determination of the scope of alternatives considered depends on the facts and circumstances at the time of the recommendation, including both the nature of the retail customer and the retail customer's investment profile, and the particular Representatives or groups of Representatives that are providing the recommendations.

<sup>&</sup>lt;sup>28</sup> *Id.* at 273.

<sup>&</sup>lt;sup>29</sup> *Id*. at 283.

#### 4. Application of Care Obligation to Account Type Recommendations

Regulation Best Interest applies to recommendations by a broker-dealer of a securities account type.<sup>30</sup> Thus, the Care Obligation requires a broker-dealer to have a reasonable basis to believe that a recommendation of a securities account type (*e.g.*, brokerage or advisory, or among the types of accounts offered by the firm) is in the retail customer's best interest at the time of the recommendation and does not place the financial or other interest of the broker-dealer ahead of the interest of the retail customer.

The Commission noted that broker-dealers would need to consider various factors in determining whether a particular account is in a particular retail customer's best interest. For example, broker-dealers generally should consider (1) the services and products provided in the account (ancillary services provided in conjunction with an account type, account monitoring services, etc.); (2) the projected cost to the retail customer of the account; (3) alternative account types available; (4) the services requested by the retail customer; and (5) the retail customer's investment profile. Moreover, retail customer-specific factors, such as those identified in the definition of "Retail Customer Investment Profile," may not be applicable or available in every context, and would depend on the facts and circumstances at the time of account type recommendation.

Where the financial professional making the recommendation is dually registered (*i.e.*, a Representative of a broker-dealer and a supervised person of an investment adviser (regardless of whether the professional works for a dual-registrant, affiliated firms, or unaffiliated firms)), the financial professional would need to make this evaluation taking into consideration the spectrum of accounts offered by the financial professional (*i.e.*, both brokerage and advisory accounts, taking into account any eligibility requirements such as account minimums), and not just brokerage accounts. Similarly, where the financial professional is only registered as a Representative of a broker-dealer (regardless of whether that broker-dealer entity is a dual-registrant or affiliated with an investment adviser), he or she would need to take into consideration only the brokerage accounts available. However, even if a broker-dealer only offered brokerage accounts, the Representative would nevertheless need to have a reasonable basis to believe that the recommended account was in the best interest of the retail customer.

#### 5. Application of Care Obligation to IRA Rollovers and Related Recommendations

Regulation Best Interest also applies to recommendations to open an IRA or to roll over assets into an IRA.<sup>31</sup> Thus, the Care Obligation requires a broker-dealer to have a reasonable basis to believe that the IRA or IRA rollover is in the best interest of the retail customer at the time of the recommendation and does not place the financial or other interest of the broker-dealer ahead of the interest of the retail customer, taking into consideration the retail customer's investment profile and other relevant factors, as well as the potential risks, rewards, and costs of the IRA or IRA rollover compared to the investor's existing 401(k) account or other circumstances. When making a recommendation to open an IRA, or to roll over workplace retirement plan assets into an IRA rather than keeping assets in a previous employer's workplace retirement plan (or rolling over assets to a new employer's workplace retirement plan), broker-dealers should consider a variety of factors, the importance of which will depend on the particular retail customer's needs and circumstances.

<sup>&</sup>lt;sup>30</sup> *Id*. at 291.

<sup>&</sup>lt;sup>31</sup> *Id*. at 295.

#### C. Conflict of Interest Obligation

The Conflict of Interest Obligation creates an overarching obligation to establish written policies and procedures to identify and disclose, or eliminate, all conflicts of interest associated with a recommendation.<sup>32</sup> This Obligation also requires broker-dealers to establish policies and procedures which are reasonably designed to mitigate or eliminate certain conflicts of interest.

Specifically, the Conflict of Interest Obligation requires broker-dealers to establish, maintain, and enforce written policies and procedures reasonably designed to identify and mitigate any conflicts of interest associated with recommendations that create an incentive for a Representative of a broker-dealer to place the interest of the broker-dealer, or the Representative making the recommendation, ahead of the interest of a retail customer. Such policies and procedures must also be designed to (1) identify and disclose any material limitations placed on the securities or investment strategies involving securities that may be recommended to a retail customer and the conflicts of interest associated with those limitations, and (2) prevent such limitations and associated conflicts from causing the broker-dealer or Representative to make recommendations that place the interests of the broker-dealer or Representative and of the interests of the retail customer. Finally, broker-dealers must identify and eliminate conflicts associated with sales contests, sales quotas, bonuses, and other non-cash compensation plans that are based on the sales of specific securities or specific types of securities within a limited period of time.

#### 1. "Reasonableness" Standard for Conflicts of Interest Policies and Procedures

In imposing the requirement for written policies and procedures to address conflicts of interest, the Commission adopted a standard of reasonableness rather than a strict liability standard that would have required a review of each and every recommendation.<sup>33</sup> In addition, rather than prescribing specific policies and procedures, the Commission recognized that broker-dealers should be permitted to reasonably design policies and procedures that are tailored to their business model, in view of the structure and characteristics of their customer relationships, including the varying degrees of recommendations provided and the types of conflicts that may arise. As the Commission noted, this reasonableness standard is consistent with Commission rules and regulations in other contexts, including under the Advisers Act. The Commission also suggested that broker-dealers consider including in their supervisory and compliance programs certain components, though not required as part of the Conflicts of Interest Obligation, that may be relevant in determining the reasonableness of the written policies and procedures. Specifically, these components include: (1) policies and procedures outlining how the firm identifies conflicts, identifying such conflicts and specifying how the broker-dealer intends to address each conflict; (2) robust compliance and monitoring systems; (3) processes to escalate non-compliance for remediation; (4) procedures that designate responsibility to business line personnel for supervision of functions and persons, including determination of compensation; (5) processes for escalating conflicts of interest; (6) processes for periodic review and testing of the adequacy and effectiveness of policies and procedures; and (7) training on policies and procedures.

It will not be enough for broker-dealers to rely on compliance with current FINRA rules to satisfy their obligations under Regulation Best Interest. The Commission believes that while FINRA rules address specific conflicts of interest, Regulation Best Interest imposes a broader obligation to address conflicts of interest both at the firm level and at the Representative level.

<sup>&</sup>lt;sup>32</sup> *Id*. at 316.

<sup>&</sup>lt;sup>33</sup> *Id*. at 307.

#### 2. Conflicts of Interest – No Materiality Requirement

As proposed, Regulation Best Interest would have required broker-dealers to adopt policies and procedures to identify, disclose, mitigate, and eliminate "material" conflicts of interest associated with recommendations. However, the Commission determined to eliminate "material" as a qualifier to "conflict of interest," and instead codified the definition of a conflict of interest to mean an interest that might incline a broker-dealer – consciously or unconsciously – to make a recommendation that is not disinterested.<sup>34</sup> Although "material" was eliminated, broker-dealers are reminded of the Disclosure Obligation, which requires broker-dealers to disclose all material facts relating to conflicts of interest Obligation to recommendations. With respect to the application of the Conflicts of Interest Obligation to non-recommendation-related services and products as such services and products were already subject to general antifraud liability and specific requirements to address associated conflicts of interest.

#### 3. Identifying Conflicts of Interest

According to the Commission, having a process to identify and appropriately categorize conflicts of interest is a critical first step to ensure that broker-dealers have reasonably designed policies and procedures to address conflicts of interest in order to comply with the Conflict of Interest Obligation.<sup>35</sup> Therefore, a reasonably designed policies and procedures to identify conflicts of interest generally should:

- define such conflicts of interest in a manner that is relevant and appropriate to the broker-dealer's business (i.e., conflicts of both the broker-dealer and its Representatives), and in a way that enables employees to understand and identify those conflicts of interest;
- establish a structure for identifying the types of conflicts that the broker-dealer and its Representatives may face;
- establish a structure to identify conflicts as the broker-dealer's business evolves;
- provide for an ongoing and regular, periodic review for the identification of conflicts associated with the broker-dealer's business; and
- establish training procedures regarding the broker-dealer's conflicts of interest, including conflicts of Representatives, how to identify such conflicts of interest, as well as defining employees' roles and responsibilities with respect to identifying such conflicts of interest.

### 4. Overarching Obligation Related to Conflicts of Interest

As stated above, the first component of the Conflicts of Interest Obligation requires a broker-dealer to establish, maintain, and enforce written policies and procedures reasonably designed to identify and disclose, or eliminate, all conflicts of interest associated with recommendations covered by Regulation Best Interest. This overarching obligation reflects the Commission's view that where a broker-dealer cannot fully and fairly disclose a conflict of interest in accordance with the Disclosure Obligation, the broker-dealer should eliminate the conflict or adequately reduce the conflict such that full and fair disclosure under the Disclosure Obligation is possible. However, the Commission determined not to provide specific examples of how broker-dealers might eliminate conflicts of interest. Instead, the Commission decided to focus the rule text on specific conflicts of interest

<sup>&</sup>lt;sup>34</sup> Id. at 309-10.

<sup>&</sup>lt;sup>35</sup> *Id*. at 315.

associated with certain sales practices based on the sale of specific securities or specific types of securities.

#### 5. Mitigation of Certain Incentives to Representatives

In proposing the overarching obligation described above, the Commission recognized the importance of the brokerage model as a potentially cost-effective option for investors. The Commission also recognized that certain compensation structures and arrangements within such business model create inherent conflicts of interest but that such compensation may be appropriate in light of the time and experience necessary for Representatives to understand investments. Therefore, rather than requiring broker-dealers to eliminate the compensation structures that give rise to inherent conflicts of interest, the Commission chose instead to preserve and promote investor choice and access to product by requiring broker-dealers to adopt policies and procedures to mitigate such conflicts of interest.<sup>36</sup> The Commission was also persuaded by concerns raised that requiring broker-dealers to establish policies and procedures to mitigate all financial incentives, including any compensation, may result in broker-dealers narrowing their product offerings and compensation structures, which would not be consistent with the Commission's desire to promote investor choice and access. Therefore, the Commission stated that broker-dealers need not eliminate transaction-based compensation to comply with Regulation Best Interest.

Rather than requiring the mitigation of all firm-level financial incentives, the Commission determined to allow such conflicts to generally be addressed through disclosure. Notwithstanding this general approach, there are certain compensation structures the Commission believes should be addressed through mitigation and not disclosure because of the potential impact that certain conflicts have on recommendations to retail customers. Specifically, as set forth in the Conflicts of Interest Obligation, broker-dealers must identify and eliminate any sales contests, sales quotas, bonuses, and non-cash compensation that are based on the sale of specific securities or specific types of securities within a limited period of time. The Commission believes that this approach appropriately balances its goal of reducing the potential harm conflicts of interest may have on broker-dealers' recommendations to retail customers' access to brokerage products and services.

#### **Guidance on Covered Incentives**

The Commission clarified that the requirement to identify and mitigate any conflicts of interest that create an incentive for the Representative to place the interest of the broker-dealer or such Representative ahead of the interest of the retail customer only applies to incentives provided to the Representative, whether by the broker-dealer or third parties controlled by or associated with the broker-dealer.<sup>37</sup> The requirement would not apply to external interests of the Representative that are not within the control of or associated with the broker-dealer's business. For example, a Representative's compensation for participating in a securities transaction that is an outside business activity would not be covered by this requirement even though the broker-dealer would need to approve the transaction and maintain appropriate books and records. For a dually registered person, this requirement would generally only apply to the incentives provided to the Representative when making a recommendation of a brokerage product or service and not when making recommendations in an investment advisory capacity.

<sup>&</sup>lt;sup>36</sup> While the Commission seeks to promote access and choice to investors, certain conflicts of interest are deemed to create too strong of an incentive for a broker-dealer to make a recommendation that places the broker-dealer's interest ahead of the retail customer's interest and must, therefore, be eliminated. See infra Section II.C.7 – Elimination of Certain Conflicts of Interest.

<sup>&</sup>lt;sup>37</sup> *Id.* at 312.

The Commission generally considers the following as examples of incentives provided to a Representative that would need to be addressed under this requirement:

- compensation from the broker-dealer or from third parties, including fees and other charges for the services provided and products sold;
- employee compensation or employment incentives (e.g., incentives tied to asset accumulation that is not expressly prohibited by the Conflicts of Interest Obligation); and
- commissions or sales charges, or other fees or financial incentives, or differential or variable compensation, whether paid by the retail customer, the broker-dealer or a third party.

Each of the above examples focuses on compensation that varies depending on the advice given, such as commissions, markups/markdowns, loads, revenue sharing, and Rule 12b-1 fees.

#### Guidance on Mitigation Methods

Whether or not a broker-dealer's policies and procedures are reasonably designed to mitigate conflicts that arise from incentives provided to Representatives depends on whether they are reasonably designed to reduce the incentive for the Representative to make a recommendation that places the interests of the Representative or the broker-dealer ahead of the retail customer's interest.<sup>38</sup> In lieu of mandating specific mitigation measures or a "one-size fits all" approach, broker-dealers have the flexibility to develop and tailor reasonably designed policies and procedures that include conflict mitigation measures that are tailored to each firm's circumstances. Reasonably designed policies and procedures should include mitigation measures that depend on the nature and significance of the incentives provided to the Representative and a variety of factors related to the broker-dealer's business model, and the complexity of the security or investment strategy. For example, the Commission stated that more stringent mitigation measures may be necessary where it is apparent that the retail customer base displays less understanding of the incentives associated with particular securities or investment strategies, where the compensation is less transparent, or in situations involving complex securities or investment strategies.

While not required elements, the Commission believes that the following non-exhaustive list of practices could be used as potential mitigation methods:

- Avoiding compensation thresholds that disproportionately increase compensation through incremental increases in sales;
- Minimizing compensation incentives for employees to favor one type of account over another; or to favor one type of product over another, proprietary or preferred provider products, or comparable products sold on a principal basis, for example, by establishing differential compensation based on neutral factors;
- Eliminating compensation incentives within comparable product lines by, for example, capping the credit that a Representative may receive across mutual funds or other comparable products across providers;
- Implementing supervisory procedures to monitor recommendations that are:
  - Near compensation thresholds;
  - Near thresholds for firm recognition;

<sup>38</sup> Id. at 330-31.

fiduciarygovernanceblog.com

- o Involve higher compensating products;
- o Proprietary products or transactions in a principal capacity; or
- Involve the rollover or transfer of assets from one type of account to another or from one product class to another;
- Adjusting compensation for Representatives who fail to adequately manage conflicts of interest; and
- Limiting the types of retail customers to whom a product, transaction, or strategy may be recommend.

The Commission cautioned against broker-dealers relying on industry practice to mitigate conflicts of interest. Instead, each firm must consider the facts and circumstances surrounding the mitigation methods employed, its particular business model, and whether or not the policies and procedures were reasonably designed to reduce the impact of the incentive to prevent the incentive from causing the Representative to place the broker-dealer's or Representative's interest ahead of the retail customer's interest.

With respect to a suggestion made by the Commission in the Proposing Release that the sophistication of retail customers might require different levels of conflicts of interest protections, the Commission clarified that the suggestion was an example and not a requirement. However, the Commission pointed to the Care Obligation, which requires that a broker-dealer, among other things, have a reasonable basis to believe that the recommendation is in the best interest of the particular retail customer. Specifically, existing suitability obligations require a broker-dealer to exercise "reasonable diligence" to ascertain and consider the retail customer's investment profile, including the investor's investment sophistication. Therefore, even if a broker-dealer is able to mitigate incentive conflicts, the broker-dealer may violate Regulation Best Interest if the recommendation does not comply with the Care Obligation.

#### 6. Mitigation of Material Limitations on Recommendations to Retail Customers

The Conflict of Interest Obligation requires, among other things, that broker-dealers establish, maintain, and enforce written policies and procedures reasonably designed to: (1) identify and disclose any material limitations broker-dealers place on their securities offerings or investment strategies involving securities and any associated conflicts of interest; and (2) prevent such limitations and associated conflicts of interest from causing the broker-dealer to make recommendations that place the broker-dealer's interest ahead of the interest of the retail customer (the "Limitations Requirement").<sup>39</sup> While the Commission believes that a broker-dealer should be able to limit its product offerings, the broker-dealer should have a process in place to disclose and address negative effects of such limitations so that retail customers are aware that the broker-dealer has a limited product offering that may make an informed investment decision by the retail customer difficult. The Commission also expressed concerns that such limitations may harm retail investors if the limitations may result in recommendations that are not in the best interest of retail customers.

For purposes of the Limitations Requirement, a "material limitation" would include recommending only proprietary products, a specific asset class, or products with third-party arrangements (*e.g.*, revenue sharing). In addition, the fact that the broker-dealer recommends only products from a select group of issuers could also be a material limitation. The Commission recognized, however, that almost all broker-dealers limit their offering of securities and investment strategies to some degree. Therefore, the fact that a broker-dealer does not offer the entire possible range of securities and investment strategies would not be a material limitation. Instead, whether the limitation is material will depend on the facts and circumstances of the extent of the limitation. Accordingly, broker-dealers will have the flexibility to develop and tailor reasonably designed policies and procedures to prevent such limitations from violating the Limitation Requirement. In crafting such policies and procedures, broker-

<sup>&</sup>lt;sup>39</sup> *Id.* at 340-41.

dealers should consider establishing product review processes for products that may be recommended, including establishing procedures for identifying and mitigating conflicts of interests associated with the product, or declining to recommend a product where the firm cannot effectively mitigate the conflict, and identifying which retail customers would qualify for recommendations of particular products.

The Commission also noted that the risk that limited product offerings may result in recommendations that are not in the retail customer's best interest is also addressed through the Care Obligation and the Disclosure Obligation.

#### 7. Elimination of Certain Conflicts of Interest

The Commission determined that it was in the public interest and consistent with the protection of investors to require that broker-dealers establish, maintain, and enforce written policies and procedures reasonably designed to identify and eliminate any sales contests, sales quotas, bonuses, and non-cash compensation that are based on the sale of specific securities or specific types of securities within a limited period of time (collectively, "Elimination Requirement").<sup>40</sup> The Commission adopted the Elimination Requirement as certain sales contests and compensation structures create high-pressure situations for Representatives to engage in sales conduct contrary to the best interest of retail customers. The Commission believes that sales contests are so pervasive that they cannot be reasonably mitigated and must be eliminated in their entirety.<sup>41</sup>

#### **Business Focus Not Prohibited**

However, the Commission recognized that some broker-dealers and Representatives focus their business on the sale of certain general categories of securities, such as mutual funds, variable annuities, bonds, or equities, and that such broker-dealers may provide compensation or other incentives related to such sales. The Commission does not believe that the Elimination Requirement necessarily requires a prohibition against a broker-dealer's focus on certain categories of securities and compensating its Representatives to further such business objectives.<sup>42</sup> The Commission does not view the providing of such incentives as being prohibited under the Conflict of Interest Obligation so long as they do not create high-pressure sales environments. Specifically, the Elimination Requirement does not apply to compensation practices based on total products sold, or asset growth or accumulation, and customer satisfaction. Thus, the Elimination Requirement would not necessarily prohibit non-cash compensation arrangements relating to the sale of investment company securities permitted under FINRA Rule 2341(I)(5) since the rule requires the compensation be "based on the total production of [Representatives] with respect to all investment company securities distributed by the member."<sup>43</sup> In addition, the Elimination Requirement does not prevent firms from limiting their offerings to proprietary products or incentivizing the sale of such products through its compensation practices, so long as the incentive is not based on the sale of specific securities or types of securities within a limited period of time.

<sup>&</sup>lt;sup>40</sup> *Id*. at 347.

<sup>&</sup>lt;sup>41</sup> But see "FINRA to Tighten Sales Contest Rules but Raise Client-Gift Limit," AdvisorHub (May 9, 2016) (quoting FINRA Chief Executive Richard Ketchum for having "congratulated the securities industry for 'moving away mostly' from product-specific sales contests"[], https://advisorhub.com/finra-to-tighten-sales-contest-rules-but-raise-client-gift-limit/.

<sup>&</sup>lt;sup>42</sup> *Id.* at 352-53.

<sup>&</sup>lt;sup>43</sup> FINRA Rule 2341(I)(5)(D).

#### Employment Related Benefits Permitted

The Commission also recognized that certain production requirements may exist for other reasons, specifically to maintain a contract of employment.<sup>44</sup> The Commission did not intend to apply the Elimination Requirement to the receipt of certain employee benefits by statutory employees, as the Commission does not consider such benefits to be non-cash compensation for purposes of Regulation Best Interest. The Elimination Requirement also does not prohibit training or education meetings, including attendance at company-sponsored meetings such as annual conferences, provided such meetings are not based on the sale of specific securities or type of securities within a limited time period.

#### Other Incentives

The Commission clarified that its emphasis on prohibiting certain incentives in the Adopting Release does not mean that all other incentives are presumed to be compliant with Regulation Best Interest.<sup>45</sup> Other incentives and practices that are not expressly prohibited are permissible if the broker-dealer establishes reasonably designed policies and procedures to disclose and mitigate the incentive created, and the broker-dealer and its Representatives comply with the Care Obligation. However, where the conflicts associated with incentives and practices are too difficult to disclose and mitigate, the broker-dealer should carefully assess whether it is able to satisfy its best interest obligation in light of the identified conflict of interest and may, in certain circumstances avoid such practice entirely.

For purposes of this requirement, "non-cash compensation" means any form of compensation received in connection with the sale and distribution of specific securities or specific types of securities that is not cash compensation, including, but not limited to, merchandise, gifts and prizes, travel expenses, and meals and lodging. Non-cash compensation does not include certain employee benefits, including healthcare and retirement benefits.

#### **D.** Compliance Obligation

The Compliance Obligation requires, in addition to the policies and procedures required by the Conflict of Interest Obligation, that broker-dealers establish, maintain, and enforce written policies and procedures reasonably designed to achieve compliance with Regulation Best Interest.<sup>46</sup> The Compliance Obligation creates an affirmative obligation under the Exchange Act with respect to the rule as a whole, while providing sufficient flexibility to allow broker-dealers to establish compliance policies and procedures that accommodate a broad range of business models. The Commission stated that the Compliance Obligation is important to help ensure that broker-dealers have strong systems of controls in place to prevent violations of Regulation Best Interest, including the component Disclosure and Care Obligations, in addition to the policies and procedures required pursuant to the Conflict of Interest Obligation, and to protect the interests of retail customers.

As with the policies and procedures requirement included in the Conflict of Interest Obligation, whether policies and procedures are reasonably designed to comply with Regulation Best Interest will depend on the facts and circumstances of a given situation. As such, the Compliance Obligation does not enumerate specific requirements that broker-dealers must include in their policies and procedures as broker-dealers are too varied in their operations for rules to impose a single set of universally applicable specific required elements. Each broker- dealer when adopting policies and procedures should consider the nature of that firm's operations and how to design such policies and procedures to prevent violations from occurring, to detect violations that have occurred, and to correct promptly any violations that have occurred.

<sup>44</sup> *Id*. at 355.

<sup>&</sup>lt;sup>45</sup> *Id*. at 356.

<sup>&</sup>lt;sup>46</sup> *Id*. at 358.

A firm's compliance policies and procedures should be reasonably designed to address and be proportionate to the scope, size, and risks associated with the operations of the firm and the types of business in which the firm engages. As such, the Commission did not mandate specific requirements pursuant to the Compliance Obligation. In addition to the required policies and procedures, depending on the size and complexity of the firm, the Commission noted that a reasonably designed compliance program generally would also include controls, remediation of non-compliance, training, and periodic review and testing.

#### III. Collateral Issues

The Adopting Release more briefly addresses several issues that are important to the application of Regulation Best Interest.

#### A. Interaction with Other Standards

Regulation Best Interest applies in addition to any other obligations of laws or regulations. The Adopting Release specifically mentions the antifraud and antimanipulation provisions of the securities laws, including Section 17(a) of the Securities Act of 1933 and Sections 9, 10(b), and 15(c) of the Exchange Act and the rules thereunder.<sup>47</sup> Thus, as under existing law, a broker-dealer recommending a security generally may be liable under the antifraud provisions if it does not give honest and complete information or disclose any material adverse facts or material conflicts of interest, including any economic self-interest.<sup>48</sup>

Broker-dealers and their Representatives will also continue to be subject to the suitability requirement of FINRA Rule 2111. This rule generally requires that a FINRA member or Representative must have a reasonable basis to believe that a recommended transaction or investment strategy involving a security or securities is suitable for the customer, based on the information obtained through reasonable diligence to ascertain the customer's investment profile. FINRA guidance under Rule 2111 states that a broker's recommendations must be consistent with his customers' best interests.<sup>49</sup> The Commission reviewed Rule 2111 and drew upon it and enhanced the suitability requirement in developing Regulation Best Interest. Although FINRA officers have suggested that the adoption of Regulation Best Interest might render Rule 2111 superfluous with respect to recommendations to retail investors,<sup>50</sup> FINRA responded to the proposal of Regulation Best Interest by considering a rule change to enhance its suitability obligation and bring it closer to the proposed Commission standard.<sup>51</sup>

### B. Intent Required to Show Violation

The Commission will not have to show scienter (i.e., an intention to deceive, manipulate, or defraud) in order to establish a violation of Regulation Best Interest. However, the rule will not be subject to strict liability; the Commission generally will have to show that a rule-breaker was at least negligent.

<sup>&</sup>lt;sup>47</sup> Adopting Release, *supra* note 2, at 43 & n.80.

<sup>&</sup>lt;sup>48</sup> *Id.* at 139 n.303.

<sup>&</sup>lt;sup>49</sup> Suitability: Additional Guidance on FINRA's New Suitability Rule, FINRA Regulatory Notice 12-25 (May 2012), <u>http://www.finra.org/industry/notices/12-25</u>.

<sup>&</sup>lt;sup>50</sup> Alison Noon, *Best Interest Standard Likely End [of FINRA Suitability Rule*, Law360 (May 16, 2019), <u>https://www.law360.com/articles/1160646/best-interest-standard-likely-end-of-finra-suitability-rule</u>.

<sup>&</sup>lt;sup>51</sup> See Quantitative Suitability: FINRA Requests Comment on Proposed Amendments to the Quantitative Suitability Obligation Under FINRA Rule 2111, FINRA Regulatory Notice 18-13 (April 20, 2018), <u>http://www.finra.org/industry/notices/18-13</u>.

### C. Waivers of Regulation Best Interest

The Adopting Release states that a broker-dealer will not be able to waive compliance with Regulation Best Interest, nor can a retail customer agree to waive her protections under Regulation Best Interest. 52 The Commission explains that this statement is based on Section 29(a) of the Exchange Act, which states that any condition, stipulation, or provision binding any person to waive compliance with any provision of the Exchange Act, or of any rule or regulation thereunder, shall be void.

### D. Private Right of Action

The Commission stated in the Adopting Release that it does not believe Regulation Best Interest creates any new private right of action or right of rescission, nor does it intend such a result.<sup>53</sup> Notwithstanding the Commission's intent and guidance, there remains a possibility that courts may take Regulation Best Interest into account in applying existing private rights of action (such as the private right of action under Exchange Act Rule 10b-5), particularly in light of the required disclosure in the Form CRS relationship summary that the broker-dealer must act in the retail customer's best interest and not put its interest ahead of the customer's.

#### E. Effect on State Law

Several commenters argued that the Commission should explicitly state that Regulation Best Interest preempts, or does not preempt, state law. The Commission avoided taking a position on this point, saying only that the preemptive effect of Regulation Best Interest on any state law governing the relationship between regulated entities and their customers would be determined in future judicial proceedings based on the specific language and effect of that state law.<sup>54</sup> Among the unknown factors, the Commission said, are the final language in any proposed state legislation or regulation adopting a fiduciary or other standard for broker-dealers, and whether that language would constitute the type of law, rule, or regulation that is expressly preempted by the securities law or impliedly preempted under principles applied by courts.<sup>55</sup>

Starting on the day of the inauguration of President Trump,<sup>56</sup> a handful of states started to pass resolutions, introduce bills, initiate enforcement actions, and propose regulations, in an attempt to either impose a uniform standard of care on broker-dealers and investment advisers or, at a minimum, highlight the differences in existing standards. These states, New York, New Jersey, Nevada, Massachusetts, Maryland and Illinois, (correctly) predicted that the Trump Administration would take a long, hard look at the 2016 DOL Fiduciary Rule. With the Fifth Circuit Court of Appeals' decision to vacate the DOL Fiduciary Rule in June 2018, the states felt emboldened to craft their own approach to standards of conduct. To date, this state action suffers from various interpretive issues on scope, questions over whether they will survive a preemption challenge, as well as industry concern that a patchwork approach will prove untenable from a compliance standpoint. Immediately below, we discuss the most important state developments in terms of broker-dealer and investment adviser standards of conduct.

<sup>&</sup>lt;sup>52</sup> Adopting Release, *supra* note 2, at 44.

<sup>&</sup>lt;sup>53</sup> Id.

<sup>&</sup>lt;sup>54</sup> *Id.* at 43, 514 n.1163.

<sup>&</sup>lt;sup>55</sup> *Id.* at 514 n.1163. Any express preemption presumably would arise under Section 15(i)(1) of the Exchange Act, which prohibits states from establishing differing or additional capital, custody, margin, financial responsibility, making and keeping records, bonding, or financial or operational reporting requirements for broker-dealers.

<sup>&</sup>lt;sup>56</sup> The first iteration of New York's proposed Investment Transparency Act was introduced by Assemblyman Jeffrey Dinowitz on Jan. 20, 2017.

#### fiduciarygovernanceblog.com

Less than ten days after the adoption of Regulation Best Interest, the Massachusetts Securities Division proposed a regulation that would impose a fiduciary standard of care on broker-dealers and investment advisers when making recommendations to retail investors with respect to (1) an investment strategy; (2) the opening of, or transfer of assets to, any type of account (including recommendations to open IRA roll-over accounts); or (3) the purchase, sale or exchange of any security. Notably, the standard of care would require a broker-dealer, agent or adviser to avoid conflicts of interest, and would also require that each recommendation or advice be made without regard to the financial or any other interest of the broker-dealer, agent, adviser, any affiliated or related entity or its officers, directors, agents, employees or contractors, or any other third party. Moreover, the sale of proprietary products, principal transactions, and broad-based sales contests/quotas may be subject to a presumption of a breach of fiduciary duty. The comment period closes on July 26.<sup>57</sup>

The Massachusetts proposed regulation is quite similar to the already-introduced New Jersey uniform fiduciary regulatory proposal, which leads us to wonder if a state regulation model is emerging. As with the Massachusetts proposal, New Jersey's proposed regulation provides that there is no presumption that disclosing a conflict of interest will satisfy the fiduciary duties. Moreover, while both Massachusetts and New Jersey seemingly preserve commissions as a compensation method, the proposed regulations raise significant interpretive questions over how a broker-dealer could receive transaction-based compensation in a manner consistent with the fiduciary standard of care. New Jersey will hold a public hearing on July 17 and its comment period has been extended to July 18.<sup>58</sup>

Nevada, meanwhile, has proposed a complicated regulation that would impose a fiduciary duty standard on broker-dealers and investment advisers. The standard of care would require that recommendations be made in the best interest of the client and that broker-dealers and investment advisers not put their interests, or the interests of others, ahead of the client's. The scope of the proposal is quite broad; seemingly any information about a specific security to a specific client could be deemed to constitute investment advice, even such general information as price and historical performance (for this reason, communications by fund managers with prospective and current investors regarding their interests in the fund may also amount to fiduciary investment advice under the proposal). Broker-dealers, but not investment advisers, may qualify for the Episodic Fiduciary Duty Exemption, which would limit the fiduciary duty to the specific investment advice provided, but not, unless otherwise required by law, have an ongoing fiduciary duty towards the client. This means that, under the Episodic Fiduciary Duty Exemption, the broker-dealer or sales representative would not have an ongoing duty to keep informed about the client's financial circumstances and obligations. It is uncertain if the Nevada Securities Division will move forward with that regulation.<sup>59</sup>

Related bills in New York, New Jersey, Illinois and Maryland have been introduced but, so far, have not garnered enough votes to pass.<sup>60</sup> These bills, with the exception of Maryland, were all disclosure-based (rather than the imposition of a standard of care). The particular political dynamics of the particular state will likely dictate whether legislation and/or regulatory action will be the preferred path, rendering it difficult to predict outcomes.

<sup>&</sup>lt;sup>57</sup> For a complete analysis, please see, Massachusetts Follows in New Jersey's Footsteps by Proposing Similar Fiduciary Duty Rule Applicable to Broker-Dealers and Investment Advisers, <u>https://fiduciarygovernanceblog.com/2019/06/17/massachusetts-proposes-fiduciaryduty-rule/</u>.

<sup>&</sup>lt;sup>58</sup> For a complete analysis, please see *New Jersey Proposes Uniform Fiduciary Standard*, <u>https://fiduciarygovernanceblog.com/2019/04/16/new-jersey-proposes-uniform-fiduciary-standard/</u>.

<sup>&</sup>lt;sup>59</sup> For a complete analysis, please see *Nevada Proposes Sweeping Fiduciary Regulation*, <u>https://fiduciarygovernanceblog.com/2019/01/22/nevada-proposes-sweeping-fiduciary-regulation/</u>.</u>

<sup>&</sup>lt;sup>60</sup> For an overview, please see A Fiduciary's 2018 Retrospective (and Predictions for 2019), https://fiduciarygovernanceblog.com/2019/01/07/a-fiduciarys-2018-retrospective-and-predictions-for-2019/.

### F. Potential Legal Challenges

There has been considerable speculation as to whether there will be a court challenge to Regulation Best Interest. For example, Barbara Roper, director of investor protection at the Consumer Federation of America, stated the day before its adoption that the rule package is vulnerable to a court challenge,<sup>61</sup> and some have interpreted this as a threat to bring such a challenge. Section 25(b) of the Exchange Act authorizes review of Commission rules by a federal court of appeals. A petition initiating such a review must be filed within 60 days after the rule's promulgation.

As of this writing, it is unclear whether any challenge will in fact be brought or, if brought, what chance it will have of success. It is assumed that any such challenge would come from investor advocates rather than the broker-dealer industry, which has been supportive of the rule.

#### G. Possible Congressional Limits on Implementation

The U.S. House of Representatives has added a provision to the appropriations bill for the fiscal year ending September 30, 2020, to provide that none of the appropriated funds for that fiscal year may be used by the Commission to implement, administer, enforce, or publicize any part of the package.<sup>62</sup> The provision is not expected to be included in the Senate version of the appropriations bill.

#### IV. Record-Making and Recordkeeping

In adopting Regulation Best Interest, the Commission has amended Rules 17a-3 and 17a-4, which specify minimum requirements for records that broker-dealers must make, and how long those records and other documents must be kept, respectively. The rules require, for each retail customer to whom a recommendation is provided, that a record be kept of all information collected from and provided to the retail customer pursuant to Regulation Best Interest, as well as the identity of each Representative of a broker-dealer responsible for the account. According to the Commission, it designed the new requirements of Rule 17a-3 to provide a means by which they can demonstrate, and Commission examiners can confirm, their compliance with the new substantive requirements of Regulation Best Interest, and that broker-dealers should already be collecting much information pursuant to the FINRA suitability rule and existing Exchange Act books and records rules. However, the Commission also stated that its current books and records requirements are not sufficient to meet the Regulation Best Interest, a broker-dealer must be able to "demonstrate that it had a reasonable basis to believe that each particular recommendation made to a retail customer was in the best interest of the customer at the time of the recommendation based on the customer's investment profile and the potential risks, rewards, and costs associated with the recommendation."

### V. Steps to Comply with Regulation Best Interest

Registered broker-dealers that recommend securities transactions or investment strategies to their retail customers must comply with Regulation Best Interest on and after June 30, 2020. While most commenters requested an implementation period of 18-24 months, the Commission "determined, in light of the importance of the protections afforded by Regulation Best Interest to retail customers, that a Compliance Date of one year after effectiveness is an appropriate timeframe for firms to conduct the requisite operational changes to their systems to establish internal processes to comply with Regulation Best Interest."<sup>63</sup> It reasoned that broker-dealers are currently subject already to supervisory obligations under Section 15(b)(4)(E) of the Exchange Act and SRO rules, including the establishment of policies and procedures reasonably designed to

<sup>&</sup>lt;sup>61</sup> Mark Schoeff Jr., Investor Advocates Slam SEC Advice Reform on Eve of Reg BI Vote, Investment News (June 4, 2019), https://www.investmentnews.com/article/20190604/FREE/190609978/investor-advocates-slam-sec-advice-reform-on-eve-of-reg-bi-vote.

<sup>&</sup>lt;sup>62</sup> Financial Services and General Government Appropriations Act, 2020, § 1003, 116<sup>th</sup> Cong., 1<sup>st</sup> Sess. (2019), <u>https://www.congress.gov/116/bills/hr3351/BILLS-116hr3351eh.pdf</u>.

<sup>&</sup>lt;sup>63</sup> Adopting Release, *supra* note 2, at 372-73.

### fiduciarygovernanceblog.com

prevent and detect violations of, and to achieve compliance with, the federal securities laws and regulations, as well as applicable SRO rule. Thus, broker-dealers "would likely establish policies and procedures to comply with Regulation Best Interest pursuant to Section 15(b)(4)(E)."<sup>64</sup> To the extent broker-dealers elect to follow this path, they could simply "adjust their current systems of supervision and compliance, as opposed to creating new systems."<sup>65</sup>

While this may be true, the Commission's reasoning fails to take into account the fact that Regulation Best Interest imposes on broker-dealers for the first time substantive disclosure and conflict of interest compliance obligations. Additional time will be needed to carefully consider these new obligations and how to best comply with them. Thus, achieving compliance with Regulation Best Interest will not simply involve the adjustment of existing supervision and compliance systems. For this reason, it seems quite possible that some brokerdealers may find meeting the compliance deadline a considerable challenge. Such firms may include smaller full service brokers with limited resources, as well as much larger full service firms, regardless of their available resources. We therefore encourage broker-dealers to get the ball rolling on the Regulation Best Interest compliance process as expeditiously as possible.

How they do so, of course, will vary from firm to firm. Nonetheless, in light of the number of separate moving parts under Regulation Best Interest that ultimately must mesh together into a whole, some firms may want to consider forming a Regulation Best Interest Compliance Committee ("BI CIC") comprised of individuals from legal, compliance, senior management, marketing, product development, and other important stakeholder groups throughout their organizations. This BI CIC may then serve as a centralized mechanism for overseeing the various work streams through to successful completion.

### **Authors**



Lawrence P. Stadulis Co-Chair, Fiduciary Governance 202.419.8407 Istadulis@stradley.com



John M. Baker Counsel 202.419.8413 jbaker@stradley.com



Peter M. Hong Partner 202.419.8429 phong@stradley.com



Michael E. Schapiro Associate 202.507.5163 mschapiro@stradley.com

<sup>&</sup>lt;sup>64</sup> Id. at 358 n.809.

# SEC Issues Interpretive Release on Investment Adviser Standard of Conduct



# **Fiduciary Governance**

June 14, 2019

**Risk&Reward** fiduciarygovernanceblog.com

# SEC Issues Interpretive Release on Investment Adviser Standard of Conduct

In conjunction with the passage of Regulation BI and the adoption of Form CRS, on June 5, 2019, the Securities and Exchange Commission issued a wide-ranging Interpretive Release that "reaffirms" and "clarif[ies]" certain aspects of the fiduciary duty that an investment adviser<sup>1</sup> owes to its clients under Section 206 of the Investment Advisers Act of 1940. This Client Alert focuses on the practical implications of the SEC's views on the scope of an adviser's fiduciary duty to clients. Investment advisers registered with the SEC should take the SEC's views contained in the Interpretive Release into account as they manage their business.

\* \* \*

Overall, the SEC stated that an investment adviser's fiduciary duty is broad, applies to the entire adviser-client relationship and is made enforceable by the anti-fraud provisions of the Advisers Act. The Interpretive Release emphasizes the SEC's specific view that an adviser's fiduciary duty under the Advisers Act is comprised of both the duty of loyalty and the duty of care, which, when combined and in light of the adoption of Regulation BI,<sup>2</sup> can be characterized as requiring an investment adviser "**to act in the best interests of its clients at all times.**"

The SEC further explained that an adviser's fiduciary duty follows the contours of the relationship between the adviser and its client, who may shape their relationship by agreement, provided that there is full and fair disclosure and informed consent.

Finally, the SEC specifically warned the industry that an adviser may not waive its fiduciary duty.<sup>3</sup>

### WHAT IS THE DUTY OF CARE?

The SEC stated that the duty of care requires an adviser to provide investment advice in the best interests of its clients based on clients' objectives. The SEC provided specific guidance by which an adviser can exercise care in its relationship with its clients as follows.

#### An Adviser Must Provide Advice That Is in the Best Interests of the Client

Notably, the SEC stated that an adviser needs to provide advice that is suitable for each client. In order to provide such advice, an adviser must have a reasonable understanding of each client's objectives.

For retail clients, the SEC stated that an adviser should:

- At a minimum, make a reasonable inquiry into the client's financial situation, level of financial sophistication, investment experience and financial goals (i.e., understand the "investment profile").
- Update the client's investment profile in order to maintain a reasonable understanding of the client's objectives and adjust the advice to reflect any changed circumstances.<sup>4</sup>

For institutional clients, the SEC explained that the nature and extent of the reasonable inquiry into clients' objectives generally is shaped by the specific investment mandates from those clients.<sup>5</sup>

© 2019 Stradley Ronon Stevens & Young, LLP | www.stradley.com | Philadelphia | Washington | Chicago | New York

Information contained in this publication should not be construed as legal advice or opinion or as a substitute for the advice of counsel. The enclosed materials may have been abridged from other sources. They are provided for educational and informational purposes for the use of clients and others who may be interested in the subject matter.

## fiduciarygovernanceblog.com

For pooled vehicles, the SEC explained that the adviser would need to have a reasonable understanding of the fund's investment guidelines and objectives.

#### An Adviser Must Have a Reasonable Belief That Advice Is in the Best Interests of the Client

The SEC confirmed that the duty of care includes the requirement that the adviser have a reasonable belief that advice is in the best interests of clients, and it provided specific guidance as to how an adviser can form that reasonable belief, noting the following:

- The adviser should evaluate its advice in the context of the portfolio that it is managing for the client and the client's objectives, taking into account the nature of the client (i.e., whether the client is a retail or an institutional client).
- For high-risk products such as penny stocks or other thinly traded securities the adviser should generally apply heightened scrutiny to whether such investments fall within a retail client's risk tolerance and objectives. A similar standard applies to complex products such as leveraged or inverse exchange-traded funds (i.e., those designed primarily as short-term trading tools for sophisticated investors) to the extent these products were in the best interests of a retail client initially, they would require daily monitoring.
- The adviser should conduct a reasonable investigation into the investment so as not to base its advice on materially inaccurate or incomplete information.
- The adviser should examine the cost (including fees and compensation) associated with investment advice, as well as the product's or strategy's investment objectives, characteristics (including any special or unusual features), liquidity, risks and potential benefits, volatility, likely performance in a variety of market and economic conditions, time horizon, and cost of exit.<sup>6</sup>

Finally, the SEC explained that the duty of care applies to all investment advice provided to clients, including advice about investment strategy, engaging a subadviser and account type.<sup>7</sup>

#### **Duty to Seek Best Execution**

Consistent with its prior statements, the SEC also specifically stated that an investment adviser's duty of care includes a duty to seek best execution of a client's transactions where the adviser has the responsibility to select broker-dealers to execute client trades. In meeting this obligation, the SEC stated that an adviser must seek to obtain the execution of transactions for each of its clients "such that the client's total cost or proceeds in each transaction are the most favorable under the circumstances." In seeking to further define this obligation of care, the SEC stated:

- An adviser should seek to obtain the execution of securities transactions on behalf of a client with the goal of maximizing value for the client under the particular circumstances occurring at the time of the transaction.
- An adviser should consider "the full range and quality of a broker's services in placing brokerage including, among other things, the value of research provided as well as execution capability, commission rate, financial responsibility, and responsiveness' to the adviser." In other words, the SEC stated that the "determinative factor" is not the lowest possible commission cost "but whether the transaction represents the best qualitative execution."
- An investment adviser should "periodically and systematically" evaluate the execution of transactions it is handling for clients.

#### Duty to Provide Advice and Monitoring Over the Course of the Relationship

Finally, the SEC stated for the first time that an investment adviser's duty of care encompasses the duty to provide advice and monitoring at a frequency that is in the best interests of the client, taking into account the scope of the agreed relationship.<sup>8</sup> The SEC stated that as a general matter, an adviser's duty to monitor extends to all personalized advice it provides to the client, including, for example, in an ongoing relationship, an evaluation of whether a client's account or program type continues to be in the client's best interests.

#### WHAT IS THE DUTY OF LOYALTY?

The second part of the Interpretive Release contained the SEC's views on the duty of loyalty, which the SEC consistently has interpreted to mean that an adviser may not subordinate its clients' interests to its own. The SEC stated that in order to meet its duty of loyalty, an adviser must make full and fair disclosure to its clients of all material facts relating to the advisory relationship, including the capacity in which the firm is acting with respect to the advice provided.<sup>9</sup>

In addition, the SEC stated that an adviser must eliminate, or at least expose through full and fair disclosure, all conflicts of interest "which might incline an investment adviser – consciously or unconsciously – to render advice which was not disinterested." Finally, the SEC explained that whether disclosure is full and fair will depend on, among other things, the nature of the client, the scope of the services, and the material fact or conflict.<sup>10</sup>

To illustrate what it views as full and fair disclosure, the SEC provided guidance on the appropriate level of specificity that advisers should attain in their disclosure, and also discussed the considerations that investment advisers should take into account for disclosure regarding conflicts related to the allocation of investment opportunities among eligible clients.

#### **Specificity of Disclosure**

- The SEC explained that for disclosure to be full and fair, it should be sufficiently specific so that a client is able to understand the material fact or conflict of interest and make an informed decision about whether to provide consent.
- Moreover, the SEC explained that disclosure that an adviser "may" have a particular conflict, without more details, is not adequate if conflict actually exists.
- The SEC elaborated that it would consider the use of "may" in disclosure to be inappropriate when the conflict exists with respect to some (but not all) types or classes of clients, advice, or transactions without additional disclosure specifying the types or classes of clients, advice, or transactions with respect to which the conflict exists.<sup>11</sup>

#### **Trade Allocations**

- The SEC reaffirmed its prior position on how an adviser can meet its duty of loyalty with regard to trade allocations by noting that when an advisor allocates investment opportunities among eligible clients, the duty of loyalty requires the adviser to eliminate or at least expose through full and fair disclosure the conflicts associated with its allocation policies including how the adviser will allocate investment opportunities such that a client can provide informed consent.
- The SEC confirmed that when allocating investment opportunities, an adviser is permitted to consider the nature and objectives of the client and the scope of the relationship and need not have pro rata allocation policies or any particular method of allocation.

In responding to commenters who objected to what they saw as subjectivity in the SEC's proposed use of the term "informed" to describe a client's consent to disclosed conflict, the SEC explained that full and fair disclosure does not require advisers to make an affirmative determination that a particular client understood the disclosure and that the client's consent to the conflict of interest was informed. Rather, the SEC stated that the disclosure should be designed to put a client in a position to be able to understand and provide informed consent to the conflict of interest.<sup>12</sup>

Finally, in what can only be viewed as a warning to the industry, the SEC stated that some conflicts may be of such a nature and extent that it would be difficult to provide disclosure to clients that "adequately conveys the material facts or the nature, magnitude, and potential effect of the conflict sufficient for a client to consent to or reject it." The SEC further warned that in some instances, disclosure may not be specific enough for a client to understand whether and how the conflict could affect the advice it receives. Specifically, for retail clients in particular, the SEC stated that where there are complex or extensive conflicts, it may be difficult to provide disclosure that is sufficiently specific but also understandable. In summary, the SEC stated that when an investment adviser cannot fully and fairly disclose a conflict of interest to a

### fiduciarygovernanceblog.com

client such that the client can provide informed consent, the adviser should either eliminate the conflict or adequately mitigate (i.e., modify practices to reduce) the conflict such that full and fair disclosure and informed consent are possible.

### CONSEQUENCES OF THE ADOPTION OF THE INTERPRETIVE RELEASE

Although the industry currently is digesting the suite of regulatory changes that the SEC adopted, given that the Interpretive Release becomes effective immediately, registered investment advisers should promptly consider examining their businesses for compliance with the Interpretive Guidance. Certain of the questions that we believe registered investment advisers need to ask include:

- Does my code of ethics or other foundational governing document lay out that my business should be run to ensure that my business conducts itself as a fiduciary and in the best interests of our clients at all times?
- Do my existing advisory contracts clearly describe the contours of the relationship and the services provided, and does that the contract, along with other disclosure documents, provide for full and fair disclosure and informed consent?
- Do my advisory contracts have "hedge clauses," and if so, are those clauses consistent with the guidance contained in the Interpretive Release?
- Do I have sufficient processes to understand each client's investment profile? And do I update that profile timely to reflect changed circumstances?
- Does my firm have policies and procedures designed to provide me with a reasonable belief that the advice I provide is in my clients' best interests? Do those policies and procedures address high-risk products for my "retail" clients? Do those policies and procedures account types and dealing with rollovers, including for prospective clients? Have I addressed each of the above questions if my firm also is registered as a broker-dealer?
- Is how I define "best execution" in my standard advisory contract or Form ADV, Part 2, consistent with the definition contained in the Interpretive Release? Do I have best execution policies and procedures, and do those procedures require me to periodically and systematically evaluate execution on behalf of my clients?
- Have I adopted policies and procedures designed to ensure that I monitor my clients' accounts at an appropriate frequency and consistent with my advisory contracts?
- Do I use the word "may" or similar words appropriately within my disclosure documents?
- Do my trade allocation procedures consider the nature and objectives of each client and the scope of each relationship?

<sup>&</sup>lt;sup>1</sup> The SEC specifically noted that the Interpretive Release applies to "robo" or "automated" advisers.

<sup>&</sup>lt;sup>2</sup> For an overview of the impact of Regulation BI and Form CRS, please see SEC Adopts Regulation Best Interest, Form CRS, and Advisers Act Interpretations, Risk&Reward, June 6, 2019 (<u>https://www.stradley.com/insights/</u>publications/2019/06/risk-and-reward-june-6-2019).

<sup>&</sup>lt;sup>3</sup> The SEC specifically stated that a contract containing (i) a statement that the adviser will not act as a fiduciary, (ii) a blanket waiver of all conflicts of interest or (iii) a waiver of any specific obligation under the Advisers Act would be inconsistent with the Advisers Act, regardless of the sophistication of the client.

Moreover, in an interesting note, the SEC withdrew the Heitman Capital Management LLC SEC Staff No-Action Letter (Feb. 12, 2007) with regard to an adviser's ability to use a "hedge clause" within an advisory contract (i.e., where the contract seeks to relieve the adviser from liability for conduct as to which the client has a nonwaivable cause of action

### fiduciarygovernanceblog.com

against the adviser provided by state or federal law). Rather, the SEC stated that whether a hedge clause violates the Advisers Act's anti-fraud provisions depends on all the surrounding facts and circumstances, including the particular circumstances of the client (e.g., sophistication). However, the SEC warned that there are few (if any) circumstances in which a hedge clause in an agreement with a retail client would be consistent with those anti-fraud provisions. The SEC further noted that whether a hedge clause in an agreement with an institutional client would violate the Advisers Act's anti-fraud provisions will be determined based on the particular facts and circumstances.

<sup>4</sup> According to the SEC, the frequency with which an adviser must update a client's investment profile should be based on facts and circumstances, including whether the adviser is aware of events that have occurred that could render inaccurate or incomplete the investment profile on which the adviser currently bases its advice.

<sup>5</sup> Interestingly, the SEC did not define what it views to be a "retail" versus an "institutional" client. While reference can certainly be drawn to definitions contained in Regulation BI and elsewhere throughout the federal securities laws, this lack of definition might have been deliberate so as to allow each adviser to assess how to classify its client base based on its own facts and circumstances.

<sup>6</sup> According to the SEC, when an adviser considers similar investment products or strategies, the adviser would not satisfy its fiduciary duty by simply advising the client to invest in the lowest-cost investment product or strategy without further analysis of other factors. In that regard, the SEC stated that an adviser can recommend a higher-cost investment or strategy if the adviser reasonably concludes that there are other factors about the investment or strategy that outweigh cost and make the investment or strategy one that is in the best interests of the client, in light of that client's objectives.

<sup>7</sup> The SEC stated that advice about account type includes advice about whether to open or invest through a certain type of account (e.g., a commission-based brokerage account or a fee-based advisory account) and advice about whether to roll over assets from one account (e.g., a retirement account) into a new or existing account that the adviser or an affiliate of the adviser manages. The SEC also specifically stated that an adviser should consider all types of accounts offered by the adviser and acknowledge to a client when the account types the adviser offers are not in the client's best interests. The SEC also noted that investment advisers have a fiduciary duty to "prospective" clients under the anti-fraud provisions of the Advisers Act – our view about these SEC pronouncements is that the SEC was indicating that an investment adviser cannot make materially misleading statements or omissions to prospective clients to induce them to become a client, and investment advisers must be able to provide appropriate advice to the client before the relationship formally starts.

<sup>8</sup> The SEC indicated that an adviser and client may agree to the "scope" of the frequency of such monitoring, provided that there is full and fair disclosure and informed consent. In an important note, the SEC emphasized that the frequency of monitoring, as well as any other material facts relating to the agreed frequency, will be "a material fact relating to the advisory relationship about which an adviser must make full and fair disclosure and obtain informed consent."

<sup>9</sup> The SEC noted the importance of disclosure for dual registrants, stating that disclosure may be accomplished through a variety of means, including, among others, written disclosure at the beginning of a relationship that clearly sets forth when a dual registrant would act in an advisory capacity and how it would provide notification of any changes in capacity.

<sup>10</sup> The SEC helpfully noted that full and fair disclosure for an institutional client (including the specificity, level of detail and explanation of terminology) can differ from full and fair disclosure for a retail client because institutional clients generally have a greater capacity and more resources than do retail clients to analyze and understand complex conflicts and their ramifications.

<sup>11</sup> The SEC further explained that the use of "may" in disclosure would be inappropriate if it simply precedes a list of all possible or potential conflicts regardless of likelihood and obfuscates actual conflicts to the point that a client cannot provide informed consent. On the other hand, the SEC noted that the word "may" could be appropriately used to disclose to a client a potential conflict that does not currently exist but might reasonably present itself in the future.

## fiduciarygovernanceblog.com

<sup>12</sup> The SEC explained that a client's informed consent can be either explicit or, depending on the facts and circumstances, implicit. Importantly, the SEC indicated that such consent need not be in writing – for example, the SEC stated that an adviser could provide appropriate disclosure through a combination of Form ADV and other disclosure and the client could implicitly consent by entering into or continuing the advisory relationship. Finally, the SEC noted that it would not be consistent with an adviser's fiduciary duty to infer or accept client consent where the adviser was aware, or reasonably should have been aware, that the client did not understand the nature and import of the conflict.

## If you would like more information, contact:



Alan P. Goldberg agoldberg@stradley.com 312.964.3503



Lawrence P. Stadulis Istadulis@stradley.com 202.419.8407



Matthew R. DiClemente mdiclemente@stradley.com 215.564.8173



Sara P. Crovitz scrovitz@stradley.com 202.507.6414



David W. Grim dgrim@stradley.com 202.507.5164

# SEC Adopts Form CRS Relationship Summary Requirement

# STRADLEY RONON

# **Fiduciary Governance**

June 18, 2019

**Risk&Reward** fiduciarygovernanceblog.com

# SEC Adopts Form CRS Relationship Summary Requirement



As part of its package of new rules, forms and interpretations under both the Investment Advisers Act of 1940 and the Securities Exchange Act of 1934, the Securities and Exchange Commission has adopted a requirement that registered investment advisers and registered broker-dealers provide a brief relationship summary to retail investors on Form CRS.

This Client Alert focuses on the practical process of drafting, filing, updating and delivering Form CRS as well as legal implications for advisers and broker-dealers subject to this new requirement.

### I. Purpose

The overarching purpose of the relationship summary is "to reduce retail investor confusion in the marketplace for brokerage and investment advisory services and to assist retail investors with the process of deciding whether to engage, or to continue to engage, a particular firm or financial professional and whether to establish, or to continue to maintain, an investment advisory or brokerage relationship."<sup>1</sup>

Following the proposal of Form CRS in April 2018, the SEC's Office of the Investor Advocate engaged the RAND Corporation to conduct investor testing on the proposed relationship summary through the administration of surveys and qualitative interviews of individual participants.<sup>2</sup> Many commenters on the proposal expressed dissatisfaction with the scope and design of the study.

© 2019 Stradley Ronon Stevens & Young, LLP | www.stradley.com | Philadelphia | Washington | Chicago | New York

Information contained in this publication should not be construed as legal advice or opinion or as a substitute for the advice of counsel. The enclosed materials may have been abridged from other sources. They are provided for educational and informational purposes for the use of clients and others who may be interested in the subject matter.

### fiduciarygovernanceblog.com

For instance, one commenter stated that "the RAND Testing was not sufficiently rigorous to appropriately assess the effectiveness of Form CRS, and we strongly believe that it does not provide a reasonable basis upon which to adopt Form CRS as proposed."<sup>3</sup> As a result, commenters urged the SEC to delay its finalization of the proposed form until it could be determined that the disclosure in Form CRS and its utility to dispel investor confusion work as intended.

Despite these concerns, the SEC adopted the final rule without delay for further testing. In its release, the SEC noted that while it "considered this input" from commenters and used it to inform certain policy choices, "feedback we have received from or on behalf of retail investors through the RAND 2018 report, surveys and studies submitted by commenters, and input received at roundtables and on Feedback Forms, demonstrate that the proposed relationship summary would be useful for retail investors and provide information, *e.g.*, about services, fees and costs, and standard of care, that would help investors to make more informed choices when deciding among firms and account options."<sup>4</sup>

In the adopting release, the SEC provides for a review of the effectiveness of Form CRS to help ensure that it "fulfills its intended purpose."5 In particular, the SEC directs the staff to review a sample of relationship summaries and provide the SEC with results of this review. There are no further details regarding the scope or timing of the review, or what the SEC will do with the staff's report once it is provided.

Time will tell whether Form CRS will ultimately be effective to assist retail investors in understanding the scope of their engagement with an investment adviser, broker-dealer or dual registrant.

### **II. Presentation and Format**

In response to several commenters arguing that the proposed Form CRS was too long and incomprehensible, the SEC resolved to streamline its presentation and format of Form CRS in adopting the final rule.

Instead of the four-page relationship summary that was proposed, the summary will now be limited to two pages (or four pages for dual registrants). Unlike the proposed instructions, the final instructions do not prescribe paper size, font size and margin width, but say only that they must be "reasonable."<sup>6</sup> The instructions also indicate that registrants should include white space and implement other design features to make the relationship summary easy to read.

Some commenters to the proposed Form CRS argued that the use of prescribed language in the instructions and the "one size fits all" approach would not appropriately address the various accounts, services and models of investment advisory and brokerage firms and therefore could be misleading to retail investors. Commenters urged the SEC to allow firms to craft their own relationship summary so that it can be tailored to the different types retail investors they service. In its adopting release, the SEC explained that, while firms will still be required to respond to a list of required topics in a prescribed order, firms will now have the flexibility to generally use their own wording to respond to the items on Form CRS.<sup>7</sup>

Several commenters to the proposed Form CRS also called for "layered" disclosure in the relationship summaries, whereby additional information could be found through cross-references, embedded URLs and QR codes. Commenters argued that "layered" disclosure would shorten and simplify the

### fiduciarygovernanceblog.com

form, as well as help to alleviate the burden of having to disclose complex information that retail investors are unlikely to read. In its adopting release, the SEC states that the instructions now permit, and in some instances require, layered disclosures and otherwise encourage the use of charts, graphs, tables and other graphics to help retail investors easily digest the information.<sup>8</sup>

Additionally, in response to commenters, roundtable participants and the RAND study, which suggested that retail investors were confused with industry jargon, the final release states that in crafting their responses on Form CRS, firms will not be permitted to use legal jargon or highly technical business terms – such as "asset-based fee" and "load" – unless firms clearly explain them in plain English, even if the firm believes a reasonable retail investor would understand those terms.<sup>9</sup>

While the scope of the information required within the relationship summary has been modified in light of commenter concerns, firms are still obligated to provide accurate information and may not omit any material facts necessary to make the required disclosures, in light of the circumstances under which they were made, not misleading.<sup>10</sup> According to the SEC, the phrase, "in *light of the circumstances under which they were made,*" was added to the final instructions in order to clarify that the disclosure is intended to be a summary, that firms must still adhere to the page limit, and that the instructions encourage firms to reference or link to additional information.<sup>11</sup> Form CRS is "not intended to create a private right of action."<sup>12</sup>

As a practical matter, firms may find it difficult to adhere to these requirements in a paper-based relationship summary. In such a format, firms must ensure that all material facts and disclosures are provided, that all technical terms are fully explained, and charts, tables or other graphics intended to help investors understand the information all fit within the confines of the two or four-page summary limit. It is unclear whether it is practical, or even possible, to disclose such information in a paper-based format within these page constraints. While the SEC permits hyperlinks and QR codes to be embedded in the text of a paper-based summary, questions remain as to the likelihood that retail investors would separately access the links provided.

### III. Standards of Conduct

The RAND 2018 Report and other surveys found that retail investors struggled to understand the standard of care for broker-dealers and investment advisers after reading the proposed Form CRS, and RAND reported that survey participants especially had difficulty reconciling the information provided in the "Obligations to You" section and the "Conflicts of Interest" section. In addition, commenters noted that retail investors did not understand the differences between a "best interest" and "fiduciary" standard, with at least one commenter noting that the confusion is likely attributable to a lack of clarity regarding the standards themselves. Yet as shown in the final release

### For more information, please contact:



Sara P. Crovitz Partner 202.507.6414 scrovitz@stradley.com



John M. Baker Counsel 202.419.8413 jbaker@stradley.com



Lauren Zychowicz Associate 215.564.8757 Izychowicz@stradley.com

### fiduciarygovernanceblog.com

for Regulation Best Interest, broker-dealers and investment advisers will continue to be subject to different standards of conduct despite the SEC's efforts to harmonize the obligations.

In the final release, firms will remain subject to the requirement to describe their legal standard of conduct using prescribed wording, but the standard of conduct disclosure is provided in the conflicts of interest section.13 The modified standard of conduct disclosure now eliminates legal jargon, such as "fiduciary," and instead uses the term "best interest" across the board, to describe how broker-dealers, investment advisers, and dual registrants must act regarding their retail customers or clients when providing recommendations as a broker-dealer or when acting as an investment adviser.

This decision is notable, considering that the final form of Regulation Best Interest still does not place a "fiduciary standard" on broker-dealers, and thus one could argue that the standards of conduct are inherently different. The harmonizing of the standard of care under Form CRS, however, appears to imply that the broker-dealer standard is as high as a fiduciary standard, even though it is not defined as such. In this regard, the SEC noted that, "we believe that requiring firms – whether broker-dealers, investment advisers, or dual registrants – to use the term 'best interest' to describe their applicable standard of conduct will clarify for retail investors their firm's legal obligation in this respect, regardless of whether that obligation arises from Regulation Best Interest or an investment adviser's fiduciary duty under the Investment Advisers Act."<sup>14</sup>

The SEC states in the adopting release that the prescribed language describing the standard of conduct broker-dealers and investment advisers owe to their customers and clients is not intended to create a private right of action.<sup>15</sup> The SEC did not address the possibility that such disclosures might support claims under existing private rights of action, such as that under rule 10b-5 under the Exchange Act.

### **IV. Contents of the Relationship Summary**

The final instructions for the relationship summary require a question-and-answer format, with standardized questions serving as the headings in a prescribed order to promote consistency and comparability. The headings will be structured and machine-readable. Under the standardized headings, firms will generally use their own wording to address the required topics, although some specific disclosures are prescribed. Firms will also be required to link to additional information, which for investment advisers will be to their Form ADV Part 2A brochures or equivalent information and for broker-dealers will be to their Regulation Best Interest disclosures.

In particular, the required headings address: (i) identifying information about the firm and a link to the SEC's website; (ii) the types of client and customer relationships and services each firm offers; (iii) the fees, costs, conflicts of interest, and required standard of conduct associated with those relationships and services; (iv) whether the firm and its financial professionals currently have reportable legal or disciplinary history; and (v) how to obtain additional information about the firm.

Some of the "Key Questions to Ask" from the proposal have been integrated into the relationship summary sections as required "conversation starters." The SEC intends that retail investors can use these questions to engage in dialogue with their financial professionals about their individual circumstances.

### V. Delivery, Filing, Updating and Recordkeeping Requirements

Under the final rule, investment advisers and broker-dealers will be obligated to deliver a relationship summary to retail investors at a certain time before or during the engagement, file copies of their relationship summaries with the SEC, update the disclosures when the information becomes materially inaccurate, and communicate any changes to retail investors who are existing clients or customers.

The relationship summary requirement applies to investment advisers and broker-dealers with retail investors. If a firm does not have any retail investors, it is not required to prepare or file a relationship summary. In particular, it appears that investment advisers to institutional separate accounts, private funds and registered funds will not be required to deliver Form CRS. In addition, the adopting release states that the SEC would not consider a broker-dealer that is serving solely as a principal underwriter to a mutual fund or variable annuity or variable life insurance contract issuer to be offering services to a retail investor for this purpose, when acting in such capacity.<sup>16</sup>

### a. Retail Investor

The SEC's final rule defines a retail investor as a natural person, or the legal representative of a natural person, who seeks to receive or receives services primarily for personal, family or household purposes. The SEC interprets a "legal representative" of a natural person to cover only non-professional legal representatives (e.g., a non-professional trustee that represents the assets of a natural person and similar representatives such as executors, conservators, and persons holding a power of attorney for a natural person).<sup>17</sup> The definition excludes natural persons seeking services for commercial or business purposes (though a relationship summary is required to be delivered to those persons who might be seeking services for a mix of personal and commercial or other non-personal purposes).<sup>18</sup> The definition does not distinguish based on a net worth or other asset threshold.<sup>19</sup> In addition, in response to commenter concerns regarding whether the definition includes plan participants, the SEC clarified that the definition will include a natural person seeking to select and retain a firm to provide brokerage or advisory services for his or her own retirement account, including but not limited to IRAs and individual accounts in workplace retirement plans, such as 401(k) plans and other tax-favored retirement plans.<sup>20</sup>

In its adopting release, the SEC stated that while the final instructions adopt a definition of "retail investor" that is consistent with the definition of "retail customer," it differs to reflect differences between the relationship summary delivery requirements and the obligations of broker-dealers under Regulation Best Interest, including "that the relationship summary is required whether or not there is a recommendation and covers any prospective and existing clients and customers (i.e., a person who 'seeks to receive or receives services') of both investment advisers and broker-dealers."<sup>21</sup>

### b. Delivery

For investment advisers, the SEC requires delivery of a relationship summary before or at the time the firm enters into an investment advisory contract and is intended to generally track the initial delivery requirement for Form ADV Part 2A. However, in a change from the proposal, broker-dealers must deliver the relationship summary to each new or prospective customer who is a retail investor before or at the earliest of one of three triggers: (i) a recommendation of an account type, a securities transaction, or an investment strategy involving securities; (ii) placing an order for the retail investor;

### fiduciarygovernanceblog.com

or (iii) the opening of a brokerage account for the retail investor. Dual registrants, and affiliated broker-dealers and investment advisers that jointly offer their services, must deliver at the earlier of the initial delivery triggers for an investment adviser or a broker-dealer.

In response to commenters' concerns, the SEC also adopted rules for delivery when changes are made to an existing account that would "materially change the nature and scope" of the firm's relationship with the retail investor. For example, firms will be required to deliver the relationship summary to existing retail investors: (i) before or at the time firms open a new account that is different from the retail investor's existing account; (ii) when they recommend that the retail investor roll over assets from a retirement account; or (iii) when they recommend or provide a new service or investment outside of a formal account (*e.g.*, variable annuities or a first-time purchase of a direct-sold mutual fund through a "check and application" process).

To facilitate retail investors receiving the relationship summary as early as possible, a firm may deliver the initial relationship summary to a new or prospective client or customer in a manner that is consistent with how the retail investor requested information about the firm or financial professional (e.g., by email if information requested by email).<sup>22</sup> The SEC is not requiring that firms make the initial relationship summary available in paper format and believes that retail investors that prefer paper communications will have the opportunity to establish relationships with firms that accommodate paper delivery.<sup>23</sup> With respect to existing clients or customers, firms must comply with the SEC's electronic delivery guidance, which provides that a person who has a right to receive a document under the federal securities laws and chooses to receive it electronically, should be provided with the information in paper form whenever specifically requesting paper.<sup>24</sup>

### c. Filing

Relationship summaries will be filed by broker-dealers on Form CRS through Web CRD and by investment advisers as Form ADV Part 3 (Form CRS) through IARD, using a text-searchable format. Dual registrants will file using both IARD and Web CRD. The use of Web CRD instead of EDGAR, as proposed, is to allow broker-dealers to use a platform which, according to commenters, is more accessible and familiar to them. The summaries will be accessible via the SEC's public website, Investor.gov, in addition to each firm's website.<sup>25</sup>

### d. Updating

Broker-dealers and investment advisers must update the relationship summary and file it within 30 days whenever any information in it becomes materially inaccurate, and any changes must be communicated to existing clients or customers within 60 days (instead of 30 days as proposed). In a change from the proposal, the SEC also added a requirement that firms delivering updated relationship summaries to existing clients or customers must highlight the most recent changes by, for example, marking the revised text or including a summary of material changes.<sup>26</sup> This additional disclosure must be filed as an exhibit to the unmarked amended relationship summary (but would not be counted toward the two-page or four-page limit, as applicable).

### e. Recordkeeping

The SEC is adopting amendments to the recordkeeping and record retention requirements under Advisers Act rule 204-2 and Exchange Act rules 17a-3 and 17a-4. Pursuant to paragraph (a)(14)(i) of

fiduciarygovernanceblog.com

Advisers Act rule 204-2 as amended, investment advisers will be required to make and preserve a record of the dates that each relationship summary was given to any client or prospective client who subsequently becomes a client. New paragraph (a)(24) of Exchange Act rule 17a-3, as adopted, will require broker-dealers to create a record of the date on which each relationship summary was provided to each retail investor, including any relationship summary provided before such retail investor opens an account. In addition, paragraph (a)(14)(i) of Advisers Act rule 204-2, as amended, will require investment advisers to retain copies of each relationship summary and each amendment or revision thereto while paragraph (e)(10) of Exchange Act rule 17a-4, as amended, will require broker-dealers to maintain and preserve a copy of each version of the relationship summary as well as the records required to be made pursuant to new paragraph (a)(24) of Exchange Act rule 17a-3. The amended rules also set forth the manner in which and the period of time for which these records must be retained.

The recordkeeping requirements were adopted as proposed, despite concerns from commenters that the requirements as written would be burdensome, especially with regard to maintaining records of prospective clients and customers who do not become clients or customers. The SEC indicated that the collection of information will be used in its examination and oversight program.<sup>27</sup>

## VI. Compliance Date

In the final rule, the SEC extended the time to comply with the relationship summary requirements. In the adopting release, the SEC noted that firms that are registered, or investment advisers who have an application for registration pending, with the Commission prior to June 30, 2020, will have a period of time beginning on May 1, 2020 until June 30, 2020 to file their initial relationship summaries with the SEC. On and after June 30, 2020, newly registered broker-dealers will be required to file their relationship summary with the SEC by the date on which their registration with the SEC becomes effective, and the SEC will not accept any initial application for registration as an investment adviser that does not include a relationship summary that satisfies the requirements of Form ADV, Part 3: Form CRS.<sup>28</sup> It is not clear what the review process will be at the SEC for relationship summaries and whether that process could slow down effectiveness of newly registering investment advisers, or whether existing broker-dealers and investment advisers should anticipate comments on their relationship summaries.

Firms will be required to deliver their relationship summary to new and prospective clients and customers who are retail investors as of the date by which they are first required to electronically file their relationship summary with the SEC. In addition, firms will be required, as part of the transition, to deliver their relationship summaries to all existing clients and customers who are retail investors on an initial one-time basis within 30 days after the date the firm is first required to file its relationship summary with the SEC.

<sup>&</sup>lt;sup>1</sup> Form CRS Relationship Summary; Amendments to Form ADV, Release Nos. 34-86032, IA-5247, at 5 (June 5, 2019) (footnotes omitted), https://www.sec.gov/rules/final/2019/34-86032.pdf; see also Exhibit B: Instructions to Form CRS, https://www.sec.gov/rules/final/2019/34-86032-appendix-b.pdf.

<sup>&</sup>lt;sup>2</sup> Form CRS Relationship Summary, supra Note 1, at 9.

<sup>&</sup>lt;sup>3</sup> Investment Adviser Association, Investor Testing of Form CRS Relationship Summary, at 2 (Dec. 4, 2018), https://www.sec.gov/comments/s7-08-18/s70818-4728888-176762.pdf. See also Consumer

Federation of America, File No. S7-08-18, Form CRS Relationship Summary, at 1 (Dec. 7, 2018), https://www.sec.gov/comments/s7-08-18/s70818-4734907-176790.pdf (the "inescapable conclusion" from the study is that the disclosures in the proposed Form CRS do not, as currently conceived, "actually support informed investor decision-making" and that many investors still fail to understand "key information that would help them determine whether a brokerage or advisory account would best suit their needs.").

- <sup>4</sup> Form CRS Relationship Summary, supra Note 1, at 11 12.
- <sup>5</sup> *Id.* at 28 29.

<sup>6</sup> The adopting release indicates that  $8\frac{1}{2}$ " x 11" paper size, at least an 11 point font size, and a minimum of 0.75" margins on all sides could be considered reasonable, but other parameters could also be reasonable. *Id.* at 48.

- <sup>7</sup> *Id*. at 21, 37.
- <sup>8</sup> *Id.* at 21 22.
- <sup>9</sup> *Id.* at 33.

<sup>10</sup> Instructions to Form CRS, *supra* Note 1, General Instruction 2.B.

- <sup>11</sup> Form CRS Relationship Summary, supra Note 1, at 42 (emphasis in original).
- <sup>12</sup> *Id.* at 43.
- <sup>13</sup> *Id*. at 143 44.
- <sup>14</sup> *Id.* at 153 54.
- <sup>15</sup> *Id*. at 145.
- <sup>16</sup> *Id.* at 224 25.

<sup>17</sup> *Id.* at 195. In other words, delivery of the relationship summary to regulated financial services professionals acting as representatives of a natural person is not required.

- <sup>18</sup> *Id.* at 193.
- <sup>19</sup> *Id.* at 193 94.

<sup>20</sup> *Id.* at 197. In addition, the SEC noted that participants in 401(k) plans and other workplace retirement plans will not be retail investors for purposes of the Form CRS delivery obligation when making certain ordinary plan elections that do not involve selecting or retaining a firm to provide brokerage or advisory services. *Id.* at 198.

- <sup>21</sup> *Id.* at 191.
- <sup>22</sup> *Id.* at 207 08.
- <sup>23</sup> *Id.* at 213 14.

<sup>24</sup> Id. at 208, 211 - 212; see Use of Electronic Media by Broker- Dealers, Transfer Agents, and Investment Advisers for Delivery of Information; Additional Examples Under the Securities Act of 1933, Securities Exchange Act of 1934, and Investment Company Act of 1940, Release Nos. 33–7288, 34–37182, IC–21945, IA–1562 (May 9, 1996), 61 Fed. Reg. 24644, 24646 (May 15, 1996), https://www.govinfo.gov/content/pkg/FR-1996-05-15/pdf/96-12176.pdf.

- <sup>25</sup> Form CRS Relationship Summary, supra Note 1, at 201.
- <sup>26</sup> *Id.* at 189.
- <sup>27</sup> *Id.* at 479.
- <sup>28</sup> *Id.* at 239.

# **Authors**

# **Fiduciary Governance Group**

## fiduciarygovernanceblog.com

# Authors

### Lawrence P. Stadulis

Co-Chair, Fiduciary Governance Washington, D.C. Istadulis@stradley.com 202.419.8407

#### Sara P. Crovitz

Partner Washington, D.C. scrovitz@stradley.com 202.507.6414

#### Alan P. Goldberg

Partner Chicago, IL agoldberg@stradley.com 312.964.3503

### Peter M. Hong

Partner Washington, DC phong@stradley.com 202.419.8429

#### John M. Baker

Counsel Washington, D.C. jbaker@stradley.com 202.419.8413

# **Fiduciary Governance Group**

fiduciarygovernanceblog.com

### **About Us:**

Investment committees, intermediaries and other service providers to retail and institutional investors continue to face a proliferation of policy, legislative, regulatory and litigation-driven changes to scopes of fiduciary status and ever-increasing fiduciary duties and requirements. These constantly evolving changes are occurring at the federal and state levels, resulting in overlapping and disparate compliance approaches. Financial institutions may be subject to multiple and conflicting sets of fiduciary or best interest obligations arising under federal and state law as a result of the nature of the different yet interrelated services they provide to their customers. These standards may also be vague and difficult to implement. This poses particular challenges to legal risk-mitigation efforts.

Stradley's fiduciary governance group counsels investment committees and intermediaries, such as investment advisers, banks, broker-dealers, retirement plan/IRA service providers, insurance providers and mutual fund directors, by identifying and making sense of this regulatory patchwork and helping clients understand the interplay of federal and state rules on 1) whether they owe a fiduciary or best interest duty to their customers, and 2) if so, the specific requirements that flow from such status. Moreover, the group keeps clients ahead of the curve by identifying emerging trends in the fiduciary landscape, such as environmental, social & governance (ESG) investing, through fiduciarygovernanceblog.com, webinars and written alerts. The fiduciary governance group also actively tracks the burgeoning state legislative efforts to impose fiduciary or comparable investment advice standards of care.

The fiduciary governance group leverages its technical understanding of both federal and state fiduciary rules to advise clients across regulatory schemes. The group seeks to help clients scale compliance programs under a particular regulatory regime to comply with other applicable fiduciary rules and requirements. The fiduciary governance group also helps identify the daylight between the fiduciary requirements of federal and state statutes and regulations.



© 2018 Stradley Ronon Stevens & Young, LLP | www.stradley.com | Philadelphia | Washington | Chicago | New York